UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

	FORM 10-Q		
\boxtimes	QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES 1934	ES EXCHANGE ACT OF	
	For the quarterly period ended April 2, 2011		
	OR		
	TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES 1934	ES EXCHANGE ACT OF	
	For the transition period from to		
	Commission file number: 000-30789		
	Entegris, Inc. (Exact name of registrant as specified in its charter)		
	(State or other jurisdiction of (I.R.S.	941551 Employer cation No.)	
		1821 • Code)	
	(978) 436-6500 (Registrant's telephone number, including area code)		
	[None] (Former name, former address and former fiscal year, if changed since last report)		
the p	cate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Section 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been so so 90 days. Yes No		
subn	cate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, evenitted and posted pursuant to Rule 405 of Regulation S-T (\S 232.405 of this chapter) during the preceding 12 months (strant was required to submit and post such files). Yes \square No \square		to be
	cate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a satisfies of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange		ā
Larg	ge accelerated filer \Box	Accelerated filer	\boxtimes
Non-	-accelerated filer \Box (Do not check if a smaller reporting company)	Smaller reporting company	
Indic	cate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes] No ⊠	
Indic	cate the number of shares outstanding of each of the issuer's classes of common stock, as of the latest practicable date.		
		at April 19, 2011 103_shares	

ENTEGRIS, INC. AND SUBSIDIARIES FORM 10-Q TABLE OF CONTENTS FOR THE QUARTER ENDED APRIL 2, 2011

	<u>Description</u>	Page
PART I		
Item 1.	<u>Unaudited Condensed Consolidated Financial Statements</u>	
	Condensed Consolidated Balance Sheets as of April 2, 2011 and December 31, 2010	3
	Condensed Consolidated Statements of Operations for the Three Months Ended April 2, 2011 and April 3, 2010	4
	Condensed Consolidated Statements of Equity and Comprehensive Income for the Three Months Ended April 2, 2011 and April 3, 2010	5
	Condensed Consolidated Statements of Cash Flows for the Three Months Ended April 2, 2011 and April 3, 2010	6
	Notes to Condensed Consolidated Financial Statements	7
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	11
Item 3.	Quantitative and Qualitative Disclosures about Market Risk	20
Item 4.	Controls and Procedures	20
PART II	Other Information	
Item 1.	<u>Legal Proceedings</u>	21
Item 1A.	Risk Factors	21
Item 6.	Exhibits	22

Item 1. Financial Statements

ENTEGRIS, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED BALANCE SHEETS

(Unaudited)

(In thousands, except share data)	April 2, 2011	Dece	mber 31, 2010
ASSETS			
Current assets:			
Cash and cash equivalents	\$ 142,578	\$	133,954
Trade accounts and notes receivable, net of allowance for doubtful accounts of \$994 and \$1,121	135,843		124,732
Inventories	100,447		101,043
Deferred tax assets, deferred tax charges and refundable income taxes	11,640		11,484
Assets held for sale	8,265		8,182
Other current assets	7,598		7,696
Total current assets	406,371		387,091
Property, plant and equipment, net of accumulated depreciation of \$224,553 and \$219,721	126,231		126,725
Other assets:			
Investments	7,453		7,017
Intangible assets, net	62,540		65,087
Deferred tax assets and other noncurrent tax assets	10,569		10,855
Other	4,346		4,610
Total assets	\$ 617,510	\$	601,385
LIABILITIES AND EQUITY			
Current liabilities:			
Accounts payable	39,969		34,631
Accrued payroll and related benefits	18,787		41,392
Other accrued liabilities	19,424		18,111
Deferred tax liabilities and income taxes payable	10,130		13,500
Total current liabilities	88,310		107,634
Pension benefit obligations and other liabilities	24,276		24,761
Deferred tax liabilities and other noncurrent tax liabilities	5,219		4,977
Commitments and contingent liabilities	_		_
Equity:			
Preferred stock, par value \$.01; 5,000,000 shares authorized; none issued and outstanding as of April 2, 2011 and December 31, 2010	_		_
Common stock, par value \$.01; 400,000,000 shares authorized; issued and outstanding shares: 134,327,603 and			
132,900,904	1,343		1,329
Additional paid-in capital	770,816		765,867
Retained deficit	(320,437)		(349,612)
Accumulated other comprehensive income	42,969		42,035
Total Entegris, Inc. shareholders' equity	494,691		459,619
Noncontrolling interest	5,014		4,394
Total equity	499,705		464,013
Total liabilities and equity	\$ 617,510	\$	601,385

ENTEGRIS, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS

(Unaudited)

	Three m	onths ended
(In thousands, except per share data)	April 2, 2011	April 3, 2010
Net sales	\$ 203,125	\$ 160,511
Cost of sales	114,780	87,360
Gross profit	88,345	73,151
Selling, general and administrative expenses	35,790	35,782
Engineering, research and development expenses	12,532	10,820
Amortization of intangible assets	2,689	4,272
Operating income	37,334	22,277
Interest expense, net	153	1,206
Other income, net	(428)	(293)
Income before income taxes and equity in affiliates	37,609	21,364
Income tax expense	8,273	4,809
Equity in net earnings of affiliates	(239)	(191)
Net income	29,575	16,746
Less net income attributable to noncontrolling interest	400	196
Net income attributable to Entegris, Inc.	\$ 29,175	\$ 16,550
Amounts attributable to Entegris, Inc.:		
Basic net income per common share:	\$ 0.22	\$ 0.13
Diluted net income per common share:	\$ 0.22	\$ 0.12
Weighted shares outstanding:		
Basic	133,699	130,954
Diluted	135,444	132,783

ENTEGRIS, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF EQUITY AND COMPREHENSIVE INCOME (Unaudited)

(In thousands)	Common shares outstanding	Common stock	Additional paid-in capital	Retained deficit	ot compr	mulated ther ehensive come		ntrolling erest	Total	sha com	tegris, Inc. areholders' aprehensive income	int comp	ontrolling erest's rehensive come
Balance at December 31, 2009	130,043	\$ 1,300	\$751,360	\$(433,968)	\$	27,500	\$	3,465	\$349,657				
Shares issued under stock plans	1,259	13	769	_		_		<u> </u>	782				
Share-based compensation expense	_	_	1,794	_		_		_	1,794				
Other, net of tax	_	_	_	_		94		_	94	\$	94	\$	
Foreign currency translation	_	_	_	_		549		(68)	481		549		(68)
Net income	_	_	_	16,550		_		196	16,746		16,550		196
Total comprehensive income										\$	17,193	\$	128
Balance at April 3, 2010	131,302	\$ 1,313	\$753,923	<u>\$(417,418)</u>	\$	28,143	\$	3,593	\$369,554				
	Common shares	Common	Additional paid-in capital	Retained	ot compr	mulated ther ehensive		ntrolling	m. 1	sha com	tegris, Inc. nreholders' nprehensive income	int comp	ontrolling erest's rehensive come
(In thousands)	outstanding	stock	сарнаі	deficit	IIIC	come	Inte	erest	Total				
(In thousands) Balance at December 31, 2010	outstanding 132,901	\$ 1,329	\$765,867	\$(349,612)		42,035	\$	4,394	\$464,013				
Balance at December 31, 2010	132,901	\$ 1,329	\$765,867						\$464,013				
Balance at December 31, 2010 Shares issued under stock plans	132,901	\$ 1,329	\$765,867 2,913						\$464,013 2,927				
Balance at December 31, 2010 Shares issued under stock plans Share-based compensation expense Tax benefit associated with stock	132,901	\$ 1,329	\$765,867 2,913 1,922						\$464,013 2,927 1,922	\$	78	\$	
Balance at December 31, 2010 Shares issued under stock plans Share-based compensation expense Tax benefit associated with stock plans	132,901	\$ 1,329	\$765,867 2,913 1,922			42,035			\$464,013 2,927 1,922			\$	
Balance at December 31, 2010 Shares issued under stock plans Share-based compensation expense Tax benefit associated with stock plans Other, net of tax	132,901	\$ 1,329	\$765,867 2,913 1,922			42,035 — — 78		4,394 — —	\$464,013 2,927 1,922 114 78		78	\$	
Balance at December 31, 2010 Shares issued under stock plans Share-based compensation expense Tax benefit associated with stock plans Other, net of tax Foreign currency translation	132,901	\$ 1,329	\$765,867 2,913 1,922	\$(349,612) ————————————————————————————————————		42,035 — — 78		4,394 — — — — 220	\$464,013 2,927 1,922 114 78 1,076		78 856	\$	

ENTEGRIS, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (Unaudited)

	Three mor	
(In thousands)	<u>April 2, 2011</u>	<u>April 3, 2010</u>
Operating activities:		
Net income	\$ 29,575	\$ 16,746
Adjustments to reconcile net income to net cash provided by operating activities:		
Depreciation	6,819	6,724
Amortization	2,689	4,272
Share-based compensation expense	1,922	1,794
Other	303	(2,336)
Changes in operating assets and liabilities:		
Trade accounts receivable and notes receivable	(10,130)	(12,612)
Inventories	724	(5,035)
Accounts payable and accrued liabilities	(15,585)	13,076
Other current assets	132	(773)
Income taxes payable and refundable income taxes	(3,559)	5,358
Other	(1,763)	809
Net cash provided by operating activities	11,127	28,023
Investing activities:		
Acquisition of property and equipment	(6,744)	(3,603)
Other	(510)	26
Net cash used in investing activities	(7,254)	(3,577)
Financing activities:		
Principal payments on short-term borrowings and long-term debt	_	(133,715)
Proceeds from short-term borrowings and long-term debt	_	113,288
Issuance of common stock	2,927	782
Other	114	
Net cash provided by (used in) financing activities	3,041	(19,645)
Effect of exchange rate changes on cash and cash equivalents	1,710	(248)
Increase in cash and cash equivalents	8,624	4,553
Cash and cash equivalents at beginning of period	133,954	68,700
Cash and cash equivalents at end of period	\$ 142,578	\$ 73,253

ENTEGRIS, INC. AND SUBSIDIARIES NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

1. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

Entegris is a leading provider of a wide range of products for purifying, protecting and transporting critical materials used in processing and manufacturing in the semiconductor and other high-technology industries. The condensed consolidated financial statements include the accounts of the Company and its majority-owned subsidiaries. Intercompany profits, transactions and balances have been eliminated in consolidation.

The preparation of condensed consolidated financial statements in conformity with accounting principles generally accepted in the United States requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities, particularly receivables, inventories, property, plant and equipment, and intangibles, accrued expenses and income taxes and related accounts, and disclosure of contingent assets and liabilities at the date of the condensed consolidated financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

In the opinion of the Company, the accompanying unaudited condensed consolidated financial statements contain all adjustments necessary to present fairly, in conformity with accounting principles generally accepted in the United States of America, the financial position as of April 2, 2011 and December 31, 2010, the results of operations for the three months ended April 2, 2011 and April 3, 2010, and equity and comprehensive income, and cash flows for the three months ended April 2, 2011 and April 3, 2010.

The condensed consolidated financial statements and notes are presented as permitted by Form 10-Q and do not contain certain information included in the Company's annual consolidated financial statements and notes. The information included in this Form 10-Q should be read in conjunction with Management's Discussion and Analysis and consolidated financial statements and notes thereto included in the Company's Form 10-K for the year ended December 31, 2010. The results of operations for the three months ended April 2, 2011 are not necessarily indicative of the results to be expected for the full year.

Fair Value of Financial Instruments The carrying value of cash equivalents, accounts receivable and accounts payable approximates fair value due to the short maturity of those instruments.

Recent Accounting Pronouncements

In October 2009, the FASB issued Accounting Standards Update (ASU) No. 2009-13, *Revenue Recognition* (Accounting Standards Codification (ASC) Topic 605) - *Multiple-Deliverable Revenue Arrangements*, *a consensus of the FASB Emerging Issues Task Force*. This guidance modifies the fair value requirements of ASC subtopic 605-25 *Revenue Recognition-Multiple Element Arrangements* by allowing the use of the "best estimate of selling price" for determining the selling price of a deliverable. A vendor is now required to use its best estimate of the selling price when vendor specific objective evidence or third-party evidence of the selling price cannot be determined. In addition, the residual method of allocating arrangement consideration is no longer permitted. This guidance was effective for the Company in 2011 and did not have a material effect on the Company's condensed consolidated financial statements.

Other Accounting Standards Updates issued not effective for the Company until after April 2, 2011 are not expected to have a material effect on the Company's condensed consolidated financial statements.

2. INVENTORIES

Inventories consist of the following:

(In thousands)	April 2, 2011	Decer	nber 31, 2010
Raw materials	\$ 27,888	\$	26,576
Work-in process	15,311		13,352
Finished goods ^(a)	56,489		60,453
Supplies	759		662
Total inventories	\$ 100,447	\$	101,043

a) Includes consignment inventories held by customers for \$5,590 and \$5,057 at April 2, 2011 and December 31, 2010, respectively.

3. INTANGIBLE ASSETS

Identifiable intangible assets, net of amortization, of \$62.5 million as of April 2, 2011 are being amortized over useful lives ranging from 3 to 15 years and are as follows:

	As of April 2, 2011		
(In thousands)	Gross carrying amount	Accumulated amortization	Net carrying value
Patents	\$ 19,089	\$ 17,730	\$ 1,359
Developed technology	74,988	54,232	20,756
Trademarks and trade names	17,345	9,404	7,941
Customer relationships	56,687	24,462	32,225
Other	3,211	2,952	259
	\$171,320	\$ 108,780	\$ 62,540

		As of December 31, 2010		
(In thousands)	Gross carrying amount	Accumulated amortization	Net carrying value	
Patents	\$ 19,050	\$ 17,588	\$ 1,462	
Developed technology	74,988	53,348	21,640	
Trademarks and trade names	17,287	9,112	8,175	
Customer relationships	56,647	23,135	33,512	
Other	5,977	5,679	298	
	\$173,949	\$ 108,862	\$ 65,087	

Aggregate amortization expense for the three months ended April 2, 2011 amounted to \$2.7 million. Estimated amortization expense for calendar years 2011 to 2015 and thereafter is approximately \$10.0 million, \$9.4 million, \$8.8 million, \$7.7 million, \$5.6 million, and \$23.7 million, respectively.

4. INCOME TAXES

The Company recorded income tax expense of \$8.3 million in the three months ended April 2, 2011 compared to income tax expense of \$4.8 million in the three months ended April 3, 2010. The effective tax rate was 22.0% in the 2011 period, compared to 22.5% in the 2010 period.

In 2011, the Company's effective tax rate was lower than U.S. statutory rates mainly due to the \$3.6 million decrease in the Company's U.S. deferred tax asset valuation allowance. Management concluded the Company will realize certain deferred tax assets related to current taxes payable and has thus released the allowance for a portion of U.S. deferred tax assets. The effective tax rate also benefitted from the Company's tax holiday in

Malaysia whereby, as a result of employment commitments, research and development expenditures and capital investments made by the Company, income from certain manufacturing activities in Malaysia is exempt from income taxes. The effective tax rate is also affected by lower tax rates in certain of the Company's taxable jurisdictions.

In 2010, the Company's effective tax rate was also lower than U.S. statutory rates mainly due to the \$2.8 million decrease in the Company's U.S. deferred tax asset valuation allowance. Management concluded the Company will realize certain deferred tax assets related to current taxes payable and has thus released the allowance for a portion of U.S. deferred tax assets. The effective tax rate also benefitted from the Company's tax holiday in Malaysia. The effective tax rate is also affected by lower tax rates in certain of the Company's taxable jurisdictions.

5. EARNINGS PER COMMON SHARE

The following table presents a reconciliation of the denominators used in the computation of basic and diluted earnings per common share.

	Three mor	nths ended
(In thousands)	April 2, 2011	April 3, 2010
Basic - weighted common shares outstanding	133,699	130,954
Weighted common shares assumed upon exercise of stock options and vesting of		
restricted common stock	1,745	1,829
Diluted - weighted common shares and common shares equivalent outstanding	135,444	132,783

Approximately 2.2 million and 5.7 million of the Company's stock options were excluded from the calculation of diluted earnings per share in the three months ended April 2, 2011 and April 3, 2010, respectively, because the exercise prices of the stock options were greater than the average price of the Company's common stock, and therefore their inclusion would have been antidilutive.

6. SEGMENT REPORTING

The Company has three reportable operating segments that provide unique products and services, are separately managed and have separate financial information evaluated regularly by the Company's chief operating decision maker in determining resource allocation and assessing performance.

The Company's financial reporting segments are Contamination Control Solutions (CCS), Microenvironments (ME), and Specialty Materials (SMD).

- CCS: provides a wide range of products and subsystems that purify, monitor and deliver critical liquids and gases used in the semiconductor manufacturing process.
- *ME*: provides products that protect wafers, reticles and electronic components at various stages of transport, processing and storage related to semiconductor manufacturing.
- *SMD*: provides specialized graphite components used in semiconductor equipment and offers low-temperature, plasma-enhanced chemical vapor deposition coatings of critical components of semiconductor manufacturing equipment used in various stages of the manufacturing process as well as graphite and silicon graphite for certain critical industrial markets.

Inter-segment sales are not significant. Segment profit is defined as net sales less direct segment operating expenses, excluding certain unallocated expenses, consisting mainly of general and administrative costs for the Company's human resources, finance and information technology functions as well as interest expense, and amortization of intangible assets.

Beginning in 2011, the Company changed its management structure for a particular department. The expenses of this department, consisting mainly of engineering, research and development expenses, are now included in the determination of ME's segment profit. These expenses had previously been included in the determination of SMD's segment profit. Accordingly, the Company has adjusted the corresponding items of segment information for earlier periods.

Summarized financial information for the Company's reportable segments is shown in the following table:

	Three	months ended
(In thousands)	April 2, 2011	April 3, 2010
Net sales		
CCS	\$ 132,244	\$ 100,742
ME	48,182	41,927
SMD	22,699	17,842
Total net sales	\$ 203,125	\$ 160,511

	Three mon	ths ended
(In thousands)	April 2, 2011	April 3, 2010
Segment profit		
CCS	\$ 39,760	\$ 28,234
ME	8,379	8,521
SMD	4,976	2,801
Total segment profit	\$ 53,115	\$ 39,556

The following table reconciles total segment profit to operating income:

	Three mon	Three months ended		
(In thousands)	April 2, 2011	April 3, 2010		
Total segment profit	\$ 53,115	\$ 39,556		
Amortization of intangibles	(2,689)	(4,272)		
Unallocated general and administrative expenses	(13,092)	(13,007)		
Operating income	\$ 37,334	\$ 22,277		

The following table presents amortization of intangibles for the Company's reportable segments:

	Three months en	Three months ended		
(In thousands)	April 2, 2011	April 3, 2010		
Amortization of intangibles				
CCS	\$ 1,236	\$ 2,815		
ME	145	137		
SMD	1,308	1,320		
	\$ 2,689	\$ 4,272		

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

Overview

This overview is not a complete discussion of the Company's financial condition, changes in financial condition and results of operations; it is intended merely to facilitate an understanding of the most salient aspects of its financial condition and operating performance and to provide a context for the detailed discussion and analysis that follows and must be read in its entirety in order to fully understand the Company's financial condition and results of operations.

Entegris, Inc. is a leading provider of products and services that purify, protect and transport the critical materials used in key technology-driven industries. Entegris derives most of its revenue from the sale of products and services to the semiconductor and related industries. The Company's customers consist primarily of semiconductor manufacturers, semiconductor equipment and materials suppliers as well as thin film transistor-liquid crystal display (TFT-LCD) and hard disk manufacturers, which are served through direct sales efforts, as well as sales and distribution relationships, in the United States, Asia, Europe and the Middle East.

The Company offers a diverse product portfolio which includes more than 17,000 standard and customized products that it believes provide the most comprehensive offering of contamination control solutions and microenvironment products and services to the microelectronics industry. Certain of these products are unit-driven and consumable products that rely on the level of semiconductor manufacturing activity to drive growth, while others rely on expansion of manufacturing capacity to drive growth. The Company's unit-driven and consumable products includes membrane-based liquid filters and housings, metal-based gas filters, resin-based gas purifiers, wafer shippers, disk-shipping containers and test assembly and packaging products and consumable graphite and silicon carbide components used in plasma etch, ion implant and chemical vapor deposition processes in semiconductor manufacturing. The Company's capital expense-driven products include components, systems and subsystems that use electro-mechanical, pressure differential and related technologies to permit semiconductor and other electronics manufacturing processes, and process carriers that protect the integrity of in-process wafers.

The Company's fiscal year is the calendar period ending each December 31. The Company's fiscal quarters consist of 13-week or 14-week periods that end on Saturday. The Company's fiscal quarters in 2011 end April 2, 2011, July 2, 2011, October 1, 2011 and December 31, 2011. Unaudited information for the three months ended April 2, 2011 and April 3, 2010 and the financial position as of April 2, 2011 and December 31, 2010 are included in this Quarterly Report on Form 10-Q.

Forward-Looking Statements

The information in this Management's Discussion and Analysis of Financial Condition and Results of Operations, except for the historical information, contains forward-looking statements. These statements are subject to risks and uncertainties and to the cautionary statement set forth below. These forward-looking statements could differ materially from actual results. The Company assumes no obligation to publicly release the results of any revision or updates to these forward-looking statements to reflect future events or unanticipated occurrences. This discussion and analysis should be read in conjunction with the condensed consolidated financial statements and the related notes thereto, which are included elsewhere in this report.

Key operating factors Key factors, which management believes have the largest impact on the overall results of operations of Entegris, Inc., include:

Level of sales Since a significant portion of the Company's product costs (except for raw materials, purchased components and direct labor) are
largely fixed in the short to medium term, an increase or decrease in sales affects gross profits and overall profitability significantly. Also, increases
or decreases in sales and operating profitability affect certain costs such as incentive compensation and commissions, which are highly variable in
nature. The Company's sales are subject to the effects of industry cyclicality, technological change, substantial competition, pricing pressures and
foreign currency fluctuation.

- *Variable margin on sales* The Company's variable margin on sales is determined by selling prices and the costs of manufacturing and raw materials. This is also affected by a number of factors, which include the Company's sales mix, purchase prices of raw material (especially resin and purchased components), competition, both domestic and international, direct labor costs, and the efficiency of the Company's production operations, among others
- *Fixed cost structure* Increases or decreases in sales have a large impact on profitability. There are a number of large fixed or semi-fixed cost components, which include salaries, indirect labor and benefits, facility costs, lease expense, and depreciation and amortization. It is not possible to vary these costs easily in the short term as volumes fluctuate. Thus changes in sales volumes can affect the usage and productivity of these cost components and can have a large effect on the Company's results of operations.

Overall Summary of Financial Results for the Three Months Ended April 2, 2011

For the three months ended April 2, 2011, net sales increased by \$42.6 million, or 27%, to \$203.1 million compared to \$160.5 million for the three months ended April 3, 2010. Net sales for the period represented the highest quarterly sales level ever achieved by the Company and reflected the eighth consecutive quarter of increased sales. The year-over-year sales increase included a favorable foreign currency translation effect of \$8.1 million related to the strengthening of most international currencies versus the U.S. dollar, most notably the Japanese yen, and Taiwanese dollar. Excluding this factor, net sales rose 21% in the first quarter of 2011 when compared to last year's first quarter.

Reflecting the year-over-year sales increase, the Company reported a higher gross profit, despite a lower gross margin rate. The Company's gross margin in the first quarter of 2011 was 43.5% versus 45.6% in the year-ago period. Operating costs, consisting of selling, general and administrative (SG&A) and engineering, research and development (ER&D) costs, rose 4% for first quarter 2011 when compared to first quarter 2010, slightly offsetting the increase in gross profit.

As a result of the aforementioned factors, the Company reported net earnings of \$29.2 million, or \$0.22 per diluted share, for the quarter ended April 2, 2011 compared to net earnings of \$16.6 million, or \$0.12 per diluted share, in the quarter ended April 3, 2010.

During the first quarter of 2011, the Company's operating activities provided cash flow of \$11.1 million. Cash and cash equivalents were \$142.6 million at April 2, 2011 compared with \$134.0 million at December 31, 2010. The Company had no outstanding short-term bank borrowings or long-term debt at April 2, 2011 or December 31, 2010.

Critical Accounting Policies

Management's discussion and analysis of financial condition and results of operations are based upon the Company's condensed consolidated financial statements, which have been prepared in accordance with accounting principles generally accepted in the United States. The preparation of these condensed consolidated financial statements requires the Company to make estimates, assumptions and judgments that affect the reported amounts of assets, liabilities, revenues and expenses and related disclosure of contingent assets and liabilities. Actual results may differ from these estimates under different assumptions or conditions.

The critical accounting policies affected most significantly by estimates, assumptions and judgments used in the preparation of the Company's condensed consolidated financial statements are described in Item 7 of its Annual Report on Form 10-K for the year ended December 31, 2010 filed with the Securities and Exchange Commission. On an ongoing basis, the Company evaluates the critical accounting policies used to prepare its consolidated financial statements, including, but not limited to, those related to accounts receivable-related valuation allowances, inventory valuation, impairment of long-lived assets, income taxes and share-based compensation. There have been no material changes in these aforementioned critical accounting policies.

Three Months Ended April 2, 2011 Compared to Three Months Ended April 3, 2010

The following table compares operating results with year-ago results, as a percentage of sales, for each caption.

Three Months Ended		
April 2, 2011	April 3, 2010	
100.0%	100.0%	
56.5	54.4	
43.5	45.6	
17.6	22.3	
6.2	6.7	
1.3	2.7	
18.4	13.9	
(0.1)	(0.8)	
0.2	0.2	
18.5	13.3	
4.1	3.0	
(0.1)	(0.1)	
14.6	10.4	
	April 2, 2011 100.0% 56.5 43.5 17.6 6.2 1.3 18.4 (0.1) 0.2 18.5 4.1 (0.1)	

Net sales For the three months ended April 2, 2011, net sales increased by \$42.6 million, or 27%, to \$203.1 million compared to \$160.5 million for the three months ended April 3, 2010. First-quarter sales growth reflected continued positive trends in both the Company's core semiconductor market and other markets served. Utilization rates and production levels at semiconductor fab customers remained at high levels, and capital spending in the semiconductor industry demonstrated solid improvement. Each of the Company's operating segments experienced a net sales increase.

The sales increase included a favorable foreign currency translation effect of \$8.1 million related to the year-over-year strengthening of most international currencies versus the U.S. dollar, most notably the Japanese yen and Taiwanese dollar. Excluding this factor, net sales rose approximately 21% in 2011 when compared to 2010.

Sales of unit-driven products increased 21%, while sales of capital-driven products rose 35%. Sales of unit-driven products represented 61% of total sales and capital-driven products represented 39% of total sales in the quarter ended April 2, 2011. For the first quarter of 2010 and fourth quarter of 2010 this split was 63%/37% and 62%/38%, respectively. This shift in relative demand for capital-driven products reflects increased capital spending by semiconductor customers for capacity-related products.

On a geographic basis, total sales to North America were 27%, Asia (excluding Japan) 40%, Europe 16% and Japan 17% compared to North America were 31%, Asia (excluding Japan) 38%, Europe 14% and Japan 17% in the first quarter a year ago. All regions experienced significant year-over-year sales increases. Sales in North America, Asia (excluding Japan), Europe and Japan rose 13%, 32%, 46% and 23% respectively, in the first quarter compared to a year ago, with a portion of the increase for sales outside North America related to favorable foreign currency translation effects.

On a sequential basis, revenues rose 12% from \$182.1 million in the fourth quarter of 2010. Sales of unit-driven products increased 8%, while sales of capital-driven products rose 17%, reflecting the increase in capital spending noted above. On a geographic basis, total sales to North America, Asia, Europe, and Japan increased 13%, 12%, 20% and 2%, respectively. The sequential sales increase included a favorable foreign currency translation effect of \$2.2 million related to the quarter-over-quarter strengthening of most international currencies versus the U.S. dollar, most notably the Taiwanese dollar. Excluding this factor, net sales rose 10% on a sequential quarter basis.

Gross profit The Company reported considerably higher gross profits, despite a decreased gross margin rate. Gross profit in the three months ended April 2, 2011 increased to \$88.3 million, up from \$73.2 million for the three months ended April 3, 2010. The gross margin rate for the first quarter of 2011 was 43.5% versus 45.6% for the first quarter of 2010.

The year-over-year sales increase accounted for the gross profit increase for 2011. The gross profit improvement associated with sales gains was offset partly by the lower gross margin percentage. The decrease in gross margin rate mainly reflected slightly lower factory utilization compared to a year earlier, as well as sales mix.

Selling, general and administrative expenses. Selling, general and administrative (SG&A) expenses were \$35.8 million in the three months ended April 2, 2011, essentially unchanged when compared to the comparable three-month period a year earlier. Reflecting the increase in net sales, SG&A expenses as a percent of net sales declined to 17.6% from 22.3% a year earlier. Employee costs, which make up about two-thirds of overall SG&A expenses, decreased slightly, mainly reflecting lower incentive compensation in 2011, in turn offsetting mostly nominal expense increases in other SG&A expense categories. The year-over-year increase in SG&A costs includes a foreign currency translation effect of \$1.3 million.

Engineering, research and development expenses Engineering, research and development (ER&D) expenses related to the support of current product lines and the development of new products and manufacturing technologies were \$12.5 million in the three months ended April 2, 2011 compared to the \$10.8 million reported in the year-ago period. ER&D expenses as a percent of net sales decreased to 6.2% from 6.7%, indicative of the increase in net sales. The increase in ER&D expense was due to higher employee and ER&D project costs.

Amortization of intangible assets Amortization of intangible assets was \$2.7 million in the three months ended April 2, 2011 compared to \$4.3 million in the year-ago period. The decline mainly reflected the absence of amortization expense for certain acquired developed technology and trade name assets which became fully amortized in 2010.

Interest expense, net Interest expense was \$0.2 million in the three months ended April 2, 2011 compared to interest expense of \$1.2 million in the year-ago period. The variance was mainly due to absence of outstanding debt in 2011. Interest expense in the first quarter of 2011 consisted mainly of the amortization of debt issuance costs associated with the Company's revolving credit facility.

Other income Other income was \$0.4 million in the three months ended April 2, 2011, mainly reflecting foreign currency transaction gains. Other income was \$0.3 million in the year-ago period, also mainly reflecting foreign currency transaction gains.

Income tax expense The Company recorded income tax expense of \$8.3 million in the three months ended April 2, 2011 compared to income tax expense of \$4.8 million in the three months ended April 3, 2010. The effective tax rate was 22.0% in the 2011 period, compared to 22.5% in the 2010 period.

In 2011, the Company's effective tax rate was lower than U.S. statutory rates mainly due to the \$3.6 million decrease in the Company's U.S. deferred tax asset valuation allowance. Management concluded the Company will realize certain deferred tax assets related to current taxes payable and has thus released the allowance for a portion of U.S. deferred tax assets. The effective tax rate also benefitted from the Company's tax holiday in Malaysia whereby, as a result of employment commitments, research and development expenditures and capital investments made by the Company, income from certain manufacturing activities in Malaysia is exempt from income taxes. The effective tax rate is also affected by lower tax rates in certain of the Company's taxable jurisdictions.

In 2010, the Company's effective tax rate was also lower than U.S. statutory rates mainly due to the \$2.8 million decrease in the Company's U.S. deferred tax asset valuation allowance. Management concluded the Company will realize certain deferred tax assets related to current taxes payable and has thus released the allowance for a portion of U.S. deferred tax assets. The effective tax rate also benefitted from the Company's tax holiday in Malaysia. The effective tax rate is also affected by lower tax rates in certain of the Company's taxable jurisdictions.

Net income attributable to Entegris, Inc. The Company recorded net income of \$29.2 million, or \$0.22 per diluted share, in the three-month period ended April 2, 2011 compared to net income of \$16.6 million, or \$0.12 per diluted share, in the three-month period ended April 3, 2010. The improvement mainly reflects the Company's higher net sales and corresponding increase in gross profit.

Segment Analysis

The Company reports its financial performance based on three reporting segments. The following is a discussion on the results of operations of these three business segments. See Note 6 "Segment Reporting" to the condensed consolidated financial statements for additional information on the Company's three segments.

The following table presents selected net sales and segment profit data for the Company's three segments for the five quarters ended April 2, 2011:

	Three months ended				
(In thousands)	April 2, 2011	December 31, 2010	October 2, 2010	July 3, 2010	April 3, 2010
Contamination Control Solutions					
Net sales	\$132,244	\$ 118,106	\$113,350	\$103,660	\$100,742
Segment profit	39,760	34,609	31,434	28,614	28,234
Microenvironments					
Net sales	\$ 48,182	\$ 45,772	\$ 47,383	\$ 47,403	\$ 41,927
Segment profit	8,379	7,602	11,110	11,697	8,521
Specialty Materials					
Net sales	\$ 22,699	\$ 18,222	\$ 17,497	\$ 16,512	\$ 17,842
Segment profit	4,976	2,847	2,903	2,529	2,801

Contamination Control Solutions (CCS)

For the first quarter of 2011, CCS net sales increased 31%, to \$132.2 million from \$100.7 million in the comparable period last year. The sales increase was led by improved sales of fluid components and systems and liquid filtration products. CCS reported a segment profit of \$39.8 million in the first quarter of 2011, up 41% from the \$28.2 million segment profit in the year-ago period. The increase in sales volume and the resulting improvement in gross profit primarily account for the year-to-year change in the segment's operating results. Slightly offsetting the increase in gross profit, CCS operating expenses increased 6%, mainly due to higher selling and engineering, research and development costs.

Sales were up 12% on a sequential basis from the fourth quarter of 2010, accompanied by a 15% increase in segment profit. The sales increase was led by improved sales of fluid components and systems. Demand for liquid and gas filtration products was strong throughout the quarter as well.

Microenvironments (ME)

For the first quarter of 2011, ME net sales increased 15%, to \$48.2 million from \$41.9 million in the comparable period last year. The revenue increase was led by improved sales of 200mm and 300mm wafer process products. ME reported a segment profit of \$8.4 million in the first quarter of 2011 compared to a \$8.5 million segment profit in the year-ago period. Lower gross margins, resulting from less favorable sales mix, contributed to the flat segment profit. In addition, ME operating expenses increased 7%, mainly due to higher selling and engineering, research and development costs associated with development of new 300mm wafer process products.

Sales were up 5% on a sequential basis from the fourth quarter of 2010, with nearly all the increase due to improved sales of 300mm wafer process products. Segment profit for ME improved by 10% sequentially.

Specialty Materials (SMD)

For the first quarter of 2011, SMD net sales increased 27%, to \$22.7 million, from \$17.8 million in the comparable period last year. Sales were up 25% on a sequential basis from the fourth quarter of 2010. The increases were due to the continued rebound in sales of specialty coated products and graphite-based components. SMD reported a segment profit of \$5.0 million in the first quarter of 2011, compared to a segment profit of \$2.8 million in both the first and fourth quarters of 2010.

Unallocated general and administrative expenses

Unallocated general and administrative expenses totaled \$13.1 million in the first quarter of 2011 compared to \$13.0 million in the first quarter of 2010.

Liquidity and Capital Resources

Operating activities Cash flow provided by operating activities totaled \$11.1 million in the three months ended April 2, 2011. Cash generated by operating activities in the first quarter of 2011 was primarily the result of net income adjusted for non-cash expenses (such as depreciation, amortization and share-based compensation). The net impact of changes in operating assets and liabilities, mainly reflecting an increase in accounts receivable and a decrease in accrued liabilities, absorbed \$30.2 million in operating cash flow.

Accounts receivable, net of foreign currency translation adjustments, increased by \$10.1 million in the first three months of 2011. This increase reflects the continued upturn in sales of the Company's products. The Company's days sales outstanding was 61 days compared to 63 days at the beginning of the year. Inventories at the end of the quarter decreased by \$0.7 million from December 31, 2010, after taking into account the impact of foreign currency translation adjustments and the provision for excess and obsolete inventory.

Accrued liabilities were \$21.3 million lower than reported at December 31, 2010, mainly due to the payment of fiscal year 2010 incentive compensation during the first quarter of 2011, while accounts payable rose by \$5.3 million.

Working capital at April 2, 2011 stood at \$318.1 million, up from \$279.5 million as of December 31, 2010, and included \$142.6 million in cash and cash equivalents, compared to cash and cash equivalents of \$134.0 million as of December 31, 2010.

Investing activities Cash flow used in investing activities totaled \$7.3 million in the three-month period ended April 2, 2011. Acquisition of property and equipment totaled \$6.7 million, primarily for additions related to manufacturing equipment, tooling and information systems. Under the terms of its revolving credit facility, the Company is restricted from making capital expenditures in excess of \$30.0 million in the first ten months of 2011. The Company does not anticipate that this limit on capital expenditures will have an adverse effect on the Company's operations.

Financing activities Cash provided by financing activities totaled \$3.0 million during the three-month period ended April 2, 2011, reflecting proceeds received in connection with common shares issued under the Company's employee stock purchase and stock option plans.

The Company has a revolving credit facility maturing November 1, 2011, with a revolving credit commitment of \$60.0 million, all of which is available to the Company dependent upon the Company's borrowing base, which is determined based on the Company's levels of qualifying domestic accounts receivable, inventories and value of its property, plant and equipment. As of April 2, 2011, the Company's borrowing base supported the entirety of its available revolving commitment amount of \$60.0 million. As of that date, the Company had no outstanding borrowings and \$0.5 million undrawn on outstanding letters of credit under the revolving credit facility. Through April 2, 2011, the Company was in compliance with all applicable debt covenants included in the terms of the revolving credit facility.

The Company also has a line of credit with two banks that provide for borrowings of Japanese yen for the Company's Japanese subsidiary, equivalent to an aggregate of approximately \$14.3 million. There were no outstanding borrowings under these lines of credit at April 2, 2011.

At April 2, 2011, the Company's shareholders' equity stood at \$494.7 million, up 8% from \$459.6 million at the beginning of the year. The increase reflected net earnings attributable to the Company of \$29.2 million, additional paid-in capital of \$1.9 million associated with the Company's share-based compensation expense, \$2.9 million received in connection with common shares issued under the Company's stock option and employee stock purchase plans, and cumulative translation adjustments of \$0.9 million.

As of April 2, 2011, the Company's sources of available funds were its cash and cash equivalents of \$142.6 million, funds available under its revolving credit facility and international credit facilities and cash flow generated from operations. The Company must maintain a minimum of \$10.0 million in cash and cash equivalents in the United States under the terms of its revolving credit facility agreement.

The Company believes that its cash and cash equivalents, funds available under its revolving credit facility and international credit facilities and cash flow generated from operations will be sufficient to meet its working capital and investment requirements for the next twelve months. If available liquidity is not sufficient to meet the Company's operating and debt service obligations as they come due, management will need to pursue alternative arrangements through additional equity or debt financing in order to meet the Company's cash requirements. However, there can be no assurance that any such financing would be available on commercially acceptable terms.

New Accounting Pronouncements

Recently adopted accounting pronouncements Refer to note 1 to the Company's condensed consolidated financial statements for a discussion of recently adopted accounting pronouncements.

Recently issued accounting pronouncements At this time, the Company does not anticipate that recently issued accounting guidance that has not yet been adopted will have a material impact on its condensed consolidated financial statements. Refer to note 1 to the Company's condensed consolidated financial statements for a discussion of recently issued accounting pronouncements.

Other Accounting Standards Updates issued not effective for the Company until after April 2, 2011 are not expected to have a material effect on the Company's condensed consolidated financial statements.

Non-GAAP Information The Company's consolidated financial statements are prepared in conformity with accounting principles generally accepted in the United States (GAAP).

The Company also provides certain non-GAAP financial measures as a complement to financial measures provided in accordance with GAAP in order to better assess and reflect trends affecting the Company's business and results of operations. Regulation G, "Conditions for Use of Non-GAAP Financial Measures," and other regulations under the Securities Exchange Act of 1934, as amended, define and prescribe the conditions for use of certain non-GAAP financial information. The Company provides non-GAAP financial measures of Adjusted EBITDA and Adjusted Operating Income together with related measures thereof, and non-GAAP Earnings Per Share (EPS).

Adjusted EBITDA, a non-GAAP term, is defined by the Company as net income attributable to Entegris, Inc. before (1) net income attributable to noncontrolling interest, (2) equity in net earnings of affiliates, (3) income tax expense (4) other income, net, (5) interest expense, net, (6) amortization of intangible assets and (7) depreciation. Adjusted Operating Income, another non-GAAP term, is defined by the Company as its Adjusted EBITDA plus depreciation. The Company also utilizes non-GAAP measures whereby Adjusted EBITDA and Adjusted Operating Income are each divided by the Company's net sales to derive Adjusted EBITDA Margin and Adjusted Operating Margin, respectively.

Non-GAAP EPS, a non-GAAP term, is defined by the Company as net income attributable to Entegris, Inc. before (1) amortization of intangible assets, (2) impairment of equity investments and (3) the tax effect of the other adjustments to net income attributable to Entegris, Inc.

The Company provides supplemental non-GAAP financial measures to better understand and manage its business and believes these measures provide investors and analysts additional and meaningful information for the assessment of the Company's ongoing results. Management also uses these non-GAAP measures to assist in the evaluation of the performance of its business segments and to make operating decisions.

Management believes the Company's non-GAAP measures help indicate the Company's baseline performance before certain gains, losses or other charges that may not be indicative of the Company's business or future outlook and offer a useful view of business performance in that the measures provide a more consistent means of comparing performance. The Company believes the non-GAAP measures aid investors' overall understanding of the Company's results by providing a higher degree of transparency for such items and providing a level of disclosure that will help investors understand how management plans, measures and evaluates the Company's business performance. Management believes that the inclusion of non-GAAP measures provides consistency in its financial reporting and facilitates investors' understanding of the Company's historic operating trends by providing an additional basis for comparisons to prior periods.

Management uses Adjusted EBITDA and Adjusted Operating Income to assist it in evaluations of the Company's operating performance by excluding items that management does not consider as relevant in the results of its ongoing operations. Internally, these non-GAAP measures are used by management for planning and forecasting purposes, including the preparation of internal budgets; for allocating resources to enhance financial performance; for evaluating the effectiveness of operational strategies; and for evaluating the Company's capacity to fund capital expenditures, secure financing and expand its business.

In addition, and as a consequence of the importance of these non-GAAP financial measures in managing its business, the Company's Board of Directors uses non-GAAP financial measures in the evaluation process to determine management compensation.

The Company believes that certain analysts and investors use Adjusted EBITDA, Adjusted Operating Income and non-GAAP EPS as supplemental measures to evaluate the overall operating performance of firms in the Company's industry. Additionally, lenders or potential lenders use Adjusted EBITDA measures to evaluate the Company's creditworthiness.

The presentation of non-GAAP financial measures is not meant to be considered in isolation, as a substitute for, or superior to, financial measures or information provided in accordance with GAAP. Management strongly encourages investors to review the Company's consolidated financial statements in their entirety and to not rely on any single financial measure.

Management notes that the use of non-GAAP measures has limitations:

First, non-GAAP financial measures are not standardized. Accordingly, the methodology used to produce the Company's non-GAAP financial measures is not computed under GAAP and may differ notably from the methodology used by other companies. For example, the Company's non-GAAP measure of Adjusted EBITDA may not be directly comparable to EBITDA or an adjusted EBITDA measure reported by other companies.

Second, the Company's non-GAAP financial measures exclude items such as amortization of intangibles and depreciation that are recurring. Amortization of intangibles and depreciation have been, and will continue to be for the foreseeable future, a significant recurring expense with an impact upon the Company's results of operations, notwithstanding the lack of immediate impact upon cash flows.

Third, there is no assurance the Company will not have future restructuring activities, gains or losses on sale of equity investments, charges for fair value mark-up of acquired inventory sold, accelerated write-offs of debt-issuance costs or similar items and, therefore, may need to record additional charges (or credits) associated with such items, including the tax effects thereon. The exclusion of these items from the Company's non-GAAP measures should not be construed as an implication that these costs are unusual, infrequent or non-recurring.

Management considers these limitations by providing specific information regarding the GAAP amounts excluded from these non-GAAP financial measures and evaluating these non-GAAP financial measures together with their most directly comparable financial measures calculated in accordance with GAAP. The calculations of Adjusted EBITDA, Adjusted Operating Income, and non-GAAP EPS, and reconciliations between these financial measures and their most directly comparable GAAP equivalents are presented below in the accompanying tables.

Reconciliation of GAAP Net income attributable to Entegris, Inc. to Adjusted operating income and Adjusted EBITDA

(In thousands)	April 2, 2011	April 3, 2010
Net sales	\$ 203,125	\$ 160,511
Net income attributable to Entegris, Inc.	\$ 29,175	\$ 16,550
Adjustments to net income attributable to Entegris, Inc.		
Net income attributable to noncontrolling interest	400	196
Equity in net earnings of affiliates	(239)	(191)
Income tax expense	8,273	4,809
Other income, net	(428)	(293)
Interest expense, net	153	1,206
GAAP – Operating income	37,334	22,277
Amortization of intangible assets	2,689	4,272
Adjusted operating income	40,023	26,549
Depreciation	6,819	6,724
Adjusted EBITDA	\$ 46,842	\$ 33,273
Adjusted operating margin	19.7%	16.5%
Adjusted EBITDA – as a % of net sales	23.1%	20.7%

Reconciliation of GAAP Earnings per Share to Non-GAAP Earnings per Share

(In thousands)	Ap	ril 2, 2011	Ap	ril 3, 2010
Net income attributable to the Company	\$	29,175	\$	16,550
Adjustments to net income attributable to the Company:				
Amortization of intangible assets		2,689		4,272
Tax effect of adjustments to net income attributable to the Company		(990)	_	(1,567)
Non-GAAP net income attributable to the Company	\$	30,874	\$	19,255
Diluted earnings per common share:	\$	0.22	\$	0.12
Effect of adjustments to net income attributable to the Company		0.01		0.02
Diluted non-GAAP earnings per common share:	\$	0.23	\$	0.15

Cautionary Statements This report contains "forward-looking statements" within the meaning of the Private Securities Litigation Reform Act of 1995. These forward-looking statements reflect the Company's current views with respect to future events and financial performance. The words "believe," "expect," "anticipate," "intends," "estimate," "forecast," "project," "should" and similar expressions are intended to identify these "forward-looking statements." All forecasts and projections in this report are "forward-looking statements," and are based on management's current expectations of the Company's near-term results, based on current information available pertaining to the Company. The risks which could cause actual results to differ from those contained in such "forward looking statements" include, without limit, the risks described in the Company's Annual Report on Form 10-K for the year ended December 31, 2010 under the headings "Risks Relating to our Business and Industry", "Risks Related to Our Borrowings", "Manufacturing Risks", "International Risks", and "Risks Related to Owning Our Securities" as well as in the Company's quarterly reports on Form 10-Q and current reports on Form 8-K as filed with the Securities and Exchange Commission.

Item 3. Quantitative and Qualitative Disclosures About Market Risk

Entegris' principal financial market risks are sensitivities to interest rates and foreign currency exchange rates. The Company's interest-bearing cash equivalents are subject to interest rate fluctuations. The Company's cash equivalents are instruments with maturities of three months or less. A 100 basis point change in interest rates would potentially increase or decrease annual net income by approximately \$0.9 million annually.

The cash flows and results of operations of the Company's foreign-based operations are subject to fluctuations in foreign exchange rates. The Company occasionally uses derivative financial instruments to manage the foreign currency exchange rate risks associated with its foreign-based operations. At April 2, 2011, the Company was party to forward contracts to deliver Malaysian Ringgits and Japanese Yen with notional values of \$20.0 million and \$34.4 million, respectively. At April 2, 2011, the Company was party to forward contracts to deliver U.S. dollars in exchange for Korean Won and Taiwanese Dollars with notional values of \$4.2 million and \$11.2 million, respectively.

Item 4. Controls and Procedures

(a) Evaluation of disclosure controls and procedures.

The Company's management, including the Chief Executive Officer and Chief Financial Officer, has conducted an evaluation of the effectiveness of the design and operation of the Company's disclosure controls and procedures (as defined under Rules 13a-15(e) and 15d-15(e) promulgated under the Securities Exchange Act of 1934 (the "1934 Act")) as of April 2, 2011. Based upon that evaluation, the Chief Executive Officer and Chief Financial Officer concluded that, as of that evaluation date, the Company's disclosure controls and procedures were effective to ensure that information required to be disclosed by the Company in the reports that it files or submits under the 1934 Act is (i) recorded, processed, summarized and reported within the time periods specified in applicable rules and forms of the Securities and Exchange Commission, and (ii) accumulated and communicated to its management, including its Chief Executive Officer and Chief Financial Officer, to allow timely decisions regarding required disclosure.

(b) Changes in internal control over financial reporting.

There has been no change in the Company's internal control over financial reporting during the most recently completed fiscal quarter that has materially affected, or is reasonably likely to materially affect, the Company's internal control over financial reporting.

PART II

OTHER INFORMATION

Item 1. Legal Proceedings

The Company is subject to various claims, legal actions, and complaints arising in the ordinary course of business. The Company believes the final outcome of these matters will not have a material adverse effect on its consolidated financial position or results of operations. The Company expenses legal costs as incurred. The following discussion provides information regarding certain litigation to which the Company was a party that were pending as of April 2, 2011.

As previously disclosed: (i) on March 3, 2003 the Company's predecessor, Mykrolis Corporation, filed a lawsuit against Pall Corporation in the United States District Court for the District of Massachusetts alleging infringement of two of the Company's U.S. patents by certain fluid separation systems and related assemblies used in photolithography applications manufactured and sold by Pall Corporation; (ii) on December 16, 2005 Pall Corporation filed a lawsuit against the Company in U.S. District Court for the Eastern District of New York alleging infringement of two of plaintiff's patents by one of the Company's gas filtration products and by the packaging for certain of the Company's liquid filtration products; (iii) on April 6, 2006 the Company filed a second lawsuit against Pall Corporation in the United States District Court for the District of Massachusetts alleging infringement of the Company's newly issued U.S. patent No.7,021,667 by certain filter assembly products used in photolithography applications that are manufactured and sold by Pall; (iv) on August 23, 2006 the Company filed a third lawsuit against Pall Corporation in the United States District Court for the District of Massachusetts alleging infringement of the Company's newly issued U.S. patent No. 7,037,424 by certain fluid separation modules and related separation apparatus, including the product known as the EZD-3 Filter Assembly, used in photolithography applications that are manufactured and sold by Pall; and (v) on May, 4, 2007 Pall Corporation filed a second lawsuit against the Company in the U.S. District Court for the Eastern District of New York alleging that certain of the Company's point-of-use filtration products infringe a newly issued Pall patent, as well as three older Pall patents. In each of the above lawsuits the plaintiff sought damages and/or injunctive relief against the alleged infringement.

On January 13, 2011 the Company and Pall Corporation entered into a Release, Settlement and License Agreement that effected a comprehensive settlement of all of the above patent infringement litigation pending between the companies. Among other things, the terms of this settlement permit both companies to continue to manufacture their existing product lines. In accordance with the terms of this settlement, a stipulation and proposed order of dismissal was filed with the appropriate court for each of the above lawsuits within 5 days following the above settlement date and the court in each case has entered a final order of dismissal.

Item 1A. Risk Factors

Other than the risk factors enumerated below, as of the date of this filing, there have been no material changes to the Risk Factors included in our Annual Report on Form 10-K for the year ended December 31, 2010, filed with the Securities and Exchange Commission on February 23, 2011.

While we have not, to date, encountered any material adverse consequences from the occurrence of the recent earthquake, tsunami and subsequent problems affecting nuclear power plants in Japan, these events could, in the future, have a negative impact on our supply chain, our ability to deliver products, the cost of our products, and the demand for our products. Approximately 20% of the Company's revenues have historically resulted from sales to Japanese customers. As a result of these events, we may, in the future, encounter reduced demand from our Japanese customers. In addition, even if supply is not interrupted or delayed, or demand from Japanese customers is not reduced, shortages of key items may result in price increases, which our suppliers may seek to pass on to us. In addition, our customers outside of Japan may be unable to produce finished products as a result of Japanese related supply chain disruptions. These customers might then cancel orders for our products. Any such occurrences could have a material adverse effect on our business, our results of operations and our financial condition.

Item 6. Exhibits

- 10.1 Fourth Amended and Restated Membrane Manufacturing and Supply Agreement, dated as of January 10, 2011, by and between Entegris, Inc. and Millipore Corporation.
- 10.2 Fluoropolymer Purchase and Supply Agreement, as amended as of January 1, 2011, by and between Entegris, Inc. and E.I. du Pont De Nemours and Company.
- 31.1 Certification of Chief Executive Officer Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
- 31.2 Certification of Chief Financial Officer Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
- 32.1 Certification of Chief Executive Officer and Chief Financial Officer Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.

CONFORMED COPY

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

ENTEGRIS, INC.

Date: April 21, 2011

/s/ Gregory B. Graves

Gregory B. Graves

Executive Vice President and Chief Financial Officer (on behalf of the registrant and as principal financial officer)

FOURTH AMENDED AND RESTATED MEMBRANE MANUFACTURE AND SUPPLY AGREEMENT

This Fourth Amended and Restated Membrane Manufacture and Supply Agreement (this "Agreement") is entered into effective as of January 10, 2011 (the "Effective Date"), between Millipore Corporation ("Millipore"), a Massachusetts corporation with its principal place of business at 290 Concord Road, Billerica, MA 01821, and Entegris, Inc. ("Entegris"), a Delaware corporation with its principal place of business at 3500 Lyman Boulevard, Chaska, MN 55318.

RECITALS

- 1. The Parties entered into a Third Amended and Restated Membrane Manufacture and Supply Agreement dated as of October 13, 2009 (the "Old Agreement") which among other things provided for the manufacture and supply of certain membranes that are used by and incorporated into products of both Millipore and Entegris, so as to appropriately ensure both Millipore and Entegris a continuing supply of such membranes.
- 2. The Old Agreement addressed Entegris' plans to move its operations out of Millipore's facility in an orderly manner. Both Parties desire to adjust the timing and related matters of such move and to coordinate such move without disruption to the mutual supply relationship between them.
- 3. Accordingly, the Parties wish to amend and restate certain provisions of the Old Agreement and to cancel and replace the Old Agreement as of the Effective Date with this Agreement.

NOW, THEREFORE, in consideration of the foregoing and the covenants and agreements set forth below, the parties hereto agree as follows:

1. <u>DEFINITIONS</u>

The following terms shall have the meanings assigned to them below whenever they are used in this Agreement including the Exhibits and Annexes hereto. Terms defined elsewhere in this Agreement shall have the meaning ascribed thereto at the location of their definition. Except where the context otherwise requires, words imparting the singular shall include the plural and vice versa, words denoting any gender shall include all genders and words denoting persons shall include bodies corporate and vice versa.

"Affiliated Company" of one of the parties shall mean any entity that controls, is controlled by, or is under common control with such party. As used herein, "control" means the possession, directly or indirectly, or the power to direct or cause the direction of the management and policies of such entity, whether through ownership of voting securities or other interests, by contract or otherwise.

Page 1 of 50

- **"Confidential Information"** shall have the meaning set forth in Section 13.1 hereof.
- "Entegris Core Business" shall mean: (i) the IC MANUFACTURING INDUSTRY including companies that manufacture integrated circuits, semiconductors, semiconductor chips and other microelectronics components, flat panel displays, solar cells and fiber optic cables, optical coatings, coated optical lenses and coated optical fibers; (ii) the IC OEM EQUIP & MATERIALS MFG. INDUSTRY including companies that manufacture equipment for the fabrication and processing of semiconductors and integrated circuits for sale to companies in the IC Manufacturing Industry as well as companies that integrate a number of components into subsystems sold to OEM equipment manufacturers for incorporation into semiconductor fabrication equipment, as well as companies that manufacture, process and supply liquids, gases, conductive materials and other advanced materials to the IC Manufacturing Industry and which provide products and systems to purify, monitor and control atmospheric conditions in clean room manufacturing environments of the IC Manufacturing Industry; and (iii) the IC RESEARCH LABORATORY INDUSTRY including university, governmental and commercial laboratories and research operations that research and/or develop innovations in the structure and composition of integrated circuits, the processes and materials used to manufacture integrated circuits and new forms of integrated circuits.
 - "Entegris Equipment" shall have the meaning set forth in Section 3.1.1 hereof.
 - "Entegris Permitted Persons" shall have the meaning set forth in Exhibit B.
 - "Equipment" shall mean the Entegris Equipment and the Millipore Equipment collectively.
- **"Facility Term"** shall mean the period commencing on the Effective Date and ending on March 31, 2014, or until this Agreement is terminated early in accordance with Section 6.2.
- **"Flat Sheet UPE Membranes"** shall mean rollstock UPE Membranes typically less than 300 microns in sheet thickness including both phobic and philic Membranes as produced at the Premises pursuant to the Old Agreement immediately prior to the Effective Date, or as modified as provided in this Agreement.
- "Information" shall mean business information, technical information and data, know-how, research information and data, formulae and other information, whether or not patentable or copyrightable, in written, oral, electronic or other tangible or intangible forms, stored in any medium, including studies, reports, records, books, contracts, instruments, surveys, discoveries, ideas, concepts, know-how, techniques, designs, specifications, drawings, blueprints, diagrams, models, prototypes, samples, flow charts, data, computer data, disks, diskettes, tapes, computer programs or other software, marketing plans, customer names, communications by or to attorneys (including attorney-

client privileged communications), memos and other materials prepared by attorneys or under their direction (including attorney work product), and other technical, financial, employee or business information or data.

- "Lease" shall mean the lease of the Premises as set forth in Section 2 hereof and in the Lease terms set forth in Exhibit B hereto.
- "Machines" shall have the meaning set forth in Section 7.1 hereof.
- "Machine Hourly Rates" shall have them meaning set forth in Section 7.1. hereof.
- "Membranes" shall mean UPE Membranes as well as any other membranes or materials that the parties may hereafter agree to add to this definition of Membranes.
- "Millipore Core Business" shall mean: (i) the BIOPHARM INDUSTRY including pharmaceutical/biotechnology and genetic engineering companies as well as manufacturers of cosmetics, medical devices, diagnostic products and clinical analytical products; (ii) the LAB & LIFE SCIENCE RESEARCH INDUSTRY including government, university and private research and testing analytical laboratories for proteomic, genomic, microbiological and similar research and analysis as well as for environmental research and analysis; and (iii) the FOOD & BEVERAGE INDUSTRY including companies that manufacture or process foods and beverages including dairy products, beer, wine, juice and soft drink manufacturers and bottled water companies.
 - "Millipore Equipment" shall have the meaning set forth in Section 3.1.2 hereof.
- "New Entegris Facility" shall mean the facility at a location chosen by Entegris on property not owned by Millipore to which Entegris will relocate all of its manufacturing operations from the Premises.
 - "Original Contract Date" shall mean March 31, 2001.
- "Other Flat Sheet UPE Membranes" shall mean Flat Sheet UPE Membranes other than Treated Flat Sheet UPE Membranes. Other Flat Sheet UPE Membranes include, as of the Effective Date, those Membranes listed under the heading "Other Flat Sheet UPE Membranes" in Exhibit A hereto.
 - "Other UPE Membranes" shall mean all UPE Membranes other than Treated Flat Sheet UPE Membranes.
 - "Other UPE Products" shall mean devices or other products which include Other UPE Membranes as a material or component.

Page 3 of 50

- **"PFA Hollow Fiber Membranes"** shall mean tubular PFA membranes having an outer diameter in the range of 500-1000 mm and an inner diameter of 100-500 mm.
 - "Premises" shall have the meaning set forth in Exhibit B.
 - "Releases" shall mean any purchase orders or other documents of purchase that Millipore may place with Entegris for UPE Membranes.
 - "Rent" shall have the meaning set forth in Exhibit B.
- **"Subsidiary"** of one of the parties shall mean any entity that is controlled by such party. As used herein, "control" of an entity means the possession, directly or indirectly, or the power to direct or cause the direction of the management and policies of such entity, whether through ownership of voting securities or other interests, by contract or otherwise.
- "Supplement" shall mean the schedule of supplemental terms and conditions specifying detailed provisions to implement the contractual commitments set forth in this Agreement relating to membrane manufacturing operations and membrane manufacturing process improvements which is attached to this Agreement as Exhibit C.
- "Supplied Party" shall mean a party to this Agreement that orders certain Membranes pursuant to this Agreement and to whom such Membranes are sold.
- **"Supplying Party"** shall mean a party to this Agreement that manufactures certain Membranes ordered by the other party pursuant to this Agreement and that sells such Membranes to the other party.
 - "Term" or "Term of this Agreement" shall mean the effective period of this Agreement as set forth in Section 6 hereof.
 - "Treated Entegris Membranes" shall mean Treated Flat Sheet UPE Membranes and Treated Other Entegris Membranes.
 - "Treated Entegris Products" shall mean devices or other products which include Treated Entegris Membranes as a material or component.
- **"Treated Other Entegris Membranes"** shall mean those Entegris membranes that are chemically treated using Millipore's VMF4 Line or using Millipore's patented VMF4 technology, including, as of the Effective Date, those membranes listed under the heading "Treated Other Entegris Membranes" in Exhibit A hereto.
- **"Treated Flat Sheet UPE Membranes"** shall mean Flat Sheet UPE Membranes that are chemically treated using Millipore's VMF4 Line or using Millipore's patented VMF4 technology, including, as of the Effective Date, those Membranes listed under the heading "Treated Flat Sheet UPE Membranes" in Exhibit A hereto.

"UPE Membranes" shall mean microporous membranes produced from an ultrahigh molecular weight polyethylene material by a melt cast process, as produced pursuant to the Old Agreement at the Premises immediately prior to the Effective Date, or as modified as provided in this Agreement.

"UPE Products" shall mean devices or other products which include UPE Membranes as a material or component.

1A. TRANSITION OF MANUFACTURING

1A.1 *Move*. The parties have agreed that Entegris will move its membrane manufacturing operations currently conducted in the Premises to the New Entegris Facility in an orderly manner (the "Move") during the Facility Term.

1A.2 *Timing; Notice.* Entegris will not move any equipment or operations sooner than March 2012. Entegris will provide to Millipore written notice (the "Move Notice" of the date on which Entegris plans to commence the Move (the "Move Date"), which Move Notice will be delivered no later than one year prior to the Move Date, but no later than February 28, 2013. Subject to the achievement of required safety stocks as described in Section 5.7, Entegris will complete the Move and have vacated the Premises no later than the end of the Facility Term. Promptly after initial delivery of the Move Notice, Entegris will prepare a detailed move schedule and provide it to Millipore. The parties will review and update the schedule during the Facility Term on a monthly basis as the Move approaches and then progresses. Millipore will supply its requirements for feasibility and qualification rolls in coordination with the schedule.

1A.3 *Coordination*. The parties will work together in good faith to facilitate such Move to minimize disruptions and costs to both parties.

1A.4 *Qualification*. Prior to the reduction or cessation of production of Membranes at the Premises, Entegris may establish its own new UPE annealing capability at the New Entegris Facility. Millipore will use its commercially reasonable efforts, in conjunction with Entegris, to qualify the Entegris' manufacturing process at the New Entegris Facility to enable the supply of qualified Membranes. Entegris will provide to Millipore in parallel with such qualification a reasonable quantity of rolls of UPE Membrane necessary for Millipore to conduct feasibility and qualification testing. All such feasibility and evaluation rolls provided to Millipore by Entegris will meet all existing Entegris specifications as agreed by the parties. If, despite each party's commercially reasonable efforts, Millipore is unable to qualify the Entegris UPE annealing line, Millipore reserves the right to perform the annealing process itself. In such

circumstances, Entegris shall provide to Millipore a sufficient quantity of extracted gel rolls, at appropriately adjusted prices, to enable Millipore to produce sufficient finished product to meet Entegris' obligations for such finished product.

1A.5 *No Supply Disruption*. Notwithstanding the Move, Entegris will at all times during the Term of this Agreement continue to meet its supply obligations to Millipore under this Agreement of any Membranes from a manufacturing process qualified by Millipore. It is the expectation of the parties that Entegris will produce Membranes from both facilities initially, noting that for some process operations duplicate equipment does not exist, and, after the Facility Term, if the New Entegris Facility and its processes are qualified by Millipore, solely from the New Entegris Facility.

1A.6 <u>No VMF4 Disruption</u>. Notwithstanding the Move, Millipore will at all times during the Term of this Agreement continue to meet its obligations to Entegris under this Agreement to provide hydrophilization of any Membranes on its VMF4 Line (or MML line if approved by Entegris).

1A.7 *Costs*. Entegris will bear all costs of dismantling, packaging, freight, shipping, installation, testing and requalification of all Entegris Equipment to be removed from the Premises.

1A.8 <u>Transfer to New Annealing Line</u>. If Entegris' new UPE annealing process can be qualified successfully by Millipore as provided above, then after such qualification Millipore shall have Millipore membrane processed by Entegris on the new Entegris annealing machine with pricing based on the costs associated with the new machine and forego any payment for existing Film I Annealing machine hours from Entegris. Prior to any successful qualification of such new process to Millipore's reasonable satisfaction, Millipore shall have Millipore membrane continue to be processed on the existing Film I annealing machine at the Premises with Entegris continuing to be charged the same rates and pricing then in effect. Any such shifting of production shall not result in a termination of the Agreement, which may be only be terminated in accordance with its terms.

2. <u>Lease of the Premises</u>

In order to enable Entegris to manufacture UPE Membranes, including Treated Flat Sheet UPE Membranes and Other Flat Sheet UPE Membranes, and Treated Other Entegris Membranes in the same production areas at Millipore's facility at 80 Ashby Road, Bedford MA. and/or such additional, reduced or substituted areas all as described in greater detail in Exhibit B hereto, and with the same processes as such UPE Membranes and Treated Other Entegris Membranes were manufactured prior to the Effective Date (both (i) for its own use and sale and for its sale of UPE Products and (ii) for supply of Flat Sheet UPE Membranes to Millipore as provided in this Agreement), Millipore and Entegris agree to the arrangements regarding Entegris' use of the Premises as are set forth in Exhibit B hereto for the duration of the Facility Term.

3. OWNERSHIP AND USE OF THE EQUIPMENT

3.1. *Ownership.* For purposes of clarification, the parties acknowledge and agree that:

3.1.1. All right, title and interest in and to the following major equipment currently used in the manufacture of UPE Membranes and/or Treated Other Entegris Membranes, now exists with, and is solely owned by Entegris (collectively, the "Entegris Equipment"):

Item#	Description	Current Location	Quantity
1.	Slurry Mixing Vessels	Bldg C – Mix Room	2
2.	Extrusion Line & support equipment (cranes, vents etc.)	Bldg D – 101	1
3.	NZE Extractors & support equipment (scales, vents etc.)	Bldg C – 103	2
4.	Release/Testing Equipment (porosimeter, flow stands, VBP stands, digital dimension equipment)	Bldg D-101	1
5.	Monomer Chemical Mixing Vessel (for philic Flat Sheet UPE Membranes)	Bldg C-Mix Room	1
6.	MSR Batch Extractors & support equipment (cranes, LS-15, etc.)	Bldg D-101	3
7.	Release/Testing Equipment (flow stands, VBP stands, digital dimension equipment)	Bldg F-Cell 5	1
8.	CUPE Mix/recirculation Pumps	Bldg C-Mix Room	2
9.	NZE Chiller Loop (~100 Tons)	Bldg C-Roof	1
10.	Oil Mist Collector (and duct/hood)	Bldg D-101	1
11.	7 Ton Edwards Chiller	Bldg D-outside	1
12.	Tiyoda-Serec Extractor	Bldg F-Cell 5	1
13.	Tiyoda-Serec Ext 42 Ton Chiller	Bldg C-roof	1

Page 7 of 50

3.1.2. All right, title and interest in and to any equipment other than the Entegris Equipment used in the manufacture of UPE Membranes and/or Treated Other Entegris Membranes, including the following equipment currently used in such manufacture (collectively, the "Millipore Equipment") is solely owned by Millipore:

Item#	Description	Current Location	Quantity
1.	Slurry Mix Stations; Control Modules 1-3	Bldg C -105	3
2.	Film 1 Annealing Line	Bldg, C-123	1
3.	VMF 4 Line (chemical modification)	Bldg. C-124	1
4.	Testing: Flow, Wet Time, Stability	Bldg. C-124	various
5.	Monomer Chemical Mixing Stations 4-6	Bldg. C-105	3

3.2. <u>Use of Millipore Equipment and Support</u>. At all times during the Facility Term, Millipore shall provide Entegris with access to and use of the Millipore Equipment, each as necessary for use in the manufacture of UPE Membranes or Treated Other Entegris Membranes in accordance with Article I of the Supplement. Entegris shall pay Millipore Machine Hourly Rates as set forth in Section 7.1 below with respect to the use of the Millipore Equipment. At all times during the Term of this Agreement Millipore will provide a reasonable level of operational assistance and general technical support assistance in resolving technical problems in UPE Membrane manufacture occurring at the Premises in accordance with Article I of the Supplement. Millipore shall be responsible for maintaining the Entegris Equipment in its current operational capability and condition, in each case as specified in Article I of the Supplement, unless the parties agree in writing during the Term of this Agreement to alter such maintenance responsibilities. To the extent Millipore provides extraordinary services to Entegris, such services will be charged in accordance with the rate structure specified in Section 7.3 of the Agreement.

4. MANUFACTURE OF MEMBRANES

- **4.1. Membrane Manufacturing Operations.** Millipore and Entegris agree that Membrane manufacturing operations during the Term of this Agreement shall be carried out in accordance with this Section 4 and with Article I of the Supplement.
 - **4.2. Guaranteed Capacity. 4.2.1.** At all times during the Term of this Agreement, Millipore agrees to have VMF4 Line capacity sufficient to handle projected philic Membrane volumes hereunder of up to 280,000 feet (~800 hours) per quarter. If necessary and at Millipore's discretion, this capacity can be accomplished by either moving Millipore products (i.e. products other than the Membranes covered by this Agreement) to

- Millipore's MML hydrophilization equipment or by moving philic Membranes covered by this Agreement to the MML hydrophilization equipment if mutually agreed upon with cost impact to be agreed upon in advance and provided that Membranes made on the MML hydrophilization equipment can be qualified successfully by Entegris.
- **4.2.2.** At all times during the Facility Term, Millipore agrees to have (i) Film 1 Annealing Line processing capacity sufficient to handle projected phobic Membrane volumes hereunder of up to 3,200,000 feet (~1400 hours) per year, and (ii) mix capacity sufficient to handle up to four (4) mixes of 400 pounds each on any days on which Entegris is running its extrusion process on the Premises.
- **4.2.3.** At all times during the Facility Term, Entegris agrees to generally schedule operations in alignment with Millipore's plan for shutdown periods and holidays. However, it is acknowledged and agreed that there will be times of high demand during the Facility Term where Entegris will be required to run the Entegris Equipment during these times. In such cases, Millipore and Entegris will mutually agree on a plan allowing for high utilization of the Entegris Equipment.
- **4.2.4.** Subject to any new or changed restrictions imposed by the applicable air emission permit(s), Entegris agrees to have sufficient capacity, across all relevant process steps, to meet Millipore's projected demand for phobic UPE Membranes that meet agreed specifications and qualification in an amount of guaranteed capacity per each calendar quarter during the Term of this Agreement of 300,000 feet. Through mutual written agreement, this guaranteed capacity level can be reduced to 200,000 feet for a mutually agreed upon time period should Millipore's firm and forecasted releases warrant.
- **4.2.5** The parties agree to commence good faith discussions in the first calendar half of 2012 regarding the possibility of continuing Entegris' access to Millipore's VMF4 Line capacity beyond the Term of this Agreement on terms and conditions mutually acceptable to both parties, if any.
- **Capacity Expansion.** Membrane manufacturing capacity will be reviewed in accordance with a mutually agreed upon schedule during the Term of this Agreement, but no less often than annually. Entegris shall be responsible to review and report on the capacity of the Entegris Equipment and Millipore shall be responsible to review and report on the capacity of the Millipore Equipment. The results of these reviews will be discussed and documented for reference and to provide a basis for capacity expansion, as may be appropriate and agreed by the parties.

- **Manufacturing Process Improvements.** Millipore and Entegris agree to implement mutually agreeable Membrane manufacturing process improvements.
- **4.5. End of Term Arrangements for Millipore.** Following the expiration or earlier termination of the Term of this Agreement, Millipore desires the full capabilities to manufacture (or have manufactured) UPE Membranes for its and its Affiliated Companies' use and sale, and for its and its Affiliated Companies' use in manufacturing (or having manufactured) UPE Products for sale. Accordingly, to facilitate Millipore's manufacture of UPE Membranes following the Term of this Agreement:
 - 4.5.1 (i) Entegris agrees in the event that Entegris during the Term of this Agreement acquires and has installed at the Premises new equipment in replacement of the existing Extrusion Line, or otherwise determines during the Term of this Agreement, in its sole discretion, that the Extrusion Line and/or one NZE Extractor are surplus and are to be disposed of, then Entegris agrees to grant Millipore an option to purchase, prior to or at the termination of this Agreement, at their then current book values and at such other reasonable terms as the parties may agree, such surplus Extrusion Line and/or NZE Extractor. Upon consummation of any such sale transaction, Entegris agrees to leave in their then current locations at the Premises, the subject Extrusion Line and/or NZE Extractor, as the case may be. Entegris agrees to notify Millipore as to whether it is granting Millipore such an option, at least eighteen (18) months prior to the termination of this Agreement; and
 - 4.5.2 In the event that the parties consummate a sale of any equipment deemed to be surplus by Entegris as specified above, Entegris agrees to provide Millipore with the know-how (including copies of all pertinent documentation) and a reasonable amount of transition assistance relating to the design, specifications, functionality, operation and maintenance of such equipment, or otherwise necessary or useful for Millipore to be able to continue the UPE Membrane manufacturing process immediately upon the termination of this Agreement, so as to be able to make or have made UPE Membrane in the same process and of the same quality as made and supplied under this Agreement. All Entegris transition assistance time shall be charged to Millipore at the rates per person-hour calculated in accordance with Section 7.3 below.

4.6. End of Term Arrangements for Entegris.

To facilitate Entegris' manufacture of UPE Membranes and Treated Entegris Membranes at a different location following the expiration or earlier termination of this Agreement:

- 4.6.1. Entegris shall remove and transport, at its own expense, the Entegris Equipment (subject to the consummation of any sale pursuant to any option to purchase certain items of such Entegris Equipment as specifically set forth in Subsection 4.5.1 above) from the Premises to a location of its choice within one-hundred eighty (180) days following such expiration or early termination. Entegris shall use its best efforts to avoid or minimize damage to the Premises or to any other part of Millipore's 80 Ashby Road facility from such removal, and shall promptly reimburse Millipore for its reasonable and actual costs of repairing any damage to the extent caused by Entegris or its agents or representatives in the process of removing the Entegris Equipment from the Premises or any other parts of such facility;
- 4.6.2. Millipore shall provide Entegris with know-how (including copies of all pertinent documentation) and a reasonable amount of transition assistance relating to the design, specifications, functionality, operation and maintenance of the Millipore Equipment, such that Entegris can make or have made, and operate and maintain, equipment substantially equivalent or comparable to the Millipore Equipment, or successfully outsource the functions performed by the Millipore Equipment, in the manufacture of UPE Membranes and Treated Other Entegris Membranes. All Millipore transition assistance shall be charged to Entegris at the rates per person-hour calculated in accordance with Section 7.3 below; and
- 4.6.3 Millipore agrees in the event that it determines during the Term of this Agreement, in its sole discretion, that the VMF4 Line and/or the Film 1 Line are surplus and are to be disposed of, then Millipore agrees to grant Entegris an option to purchase, prior to or at the termination of this Agreement, at their then current book values and at such other reasonable terms as the parties may agree, such surplus VMF4 Line and/or the Film 1 Line. Upon consummation of any such sale transaction, Entegris shall remove and transport, at its own expense, any such purchased surplus equipment from the Premises to a location of its choice within one-hundred eighty (180) days following such expiration or early termination. Entegris shall use its best efforts to avoid or minimize damage to the Premises or to any other part of Millipore's 80 Ashby Road facility from such removal, and shall promptly reimburse Millipore for its reasonable and actual costs of repairing any damage to the extent caused by Entegris or its agents or representatives in the process of removing such purchased surplus equipment from the Premises or any other parts of such facility. Millipore agrees to notify Entegris as to whether it is granting Entegris such an option, at least eighteen (18) months prior to the termination of this Agreement

- **4.7. Joint Know-How.** In the event that any know-how results from or is developed in the course of the manufacture of UPE Membranes or Treated Other Entegris Membranes in the Premises during the Facility Term of this Agreement (including the use of Millipore Equipment in such manufacture), whether by employees of Millipore, employees of Entegris or jointly, such know-how shall be jointly owned by Entegris and Millipore. Millipore shall have rights to use such know-how in all fields other than the Entegris Core Business, and Entegris shall have rights to use such know-how in all fields other than the Millipore Core Business.
- **Additional Membranes**. Entegris shall have the right to add other membranes to the list and definition of "UPE Membranes" during the Term of this Agreement, subject to (i) Millipore's approval (on grounds of safety, compliance with laws, or avoidance of damage to the Millipore Equipment, the Premises or any other parts of Millipore's 80 Ashby Road facility) of the manufacture of such additional UPE Membranes, which approval shall not be unreasonably withheld or delayed, and (ii) Millipore's having sufficient space and equipment capacity for such additional manufacture, and (iii) scheduling of use of the Millipore Equipment as shall be negotiated by the parties in good faith.
- **4.9.** <u>Additional Equipment</u>. Subject to Millipore's prior approval, which approval shall not be unreasonably withheld or delayed, Entegris shall have the right to locate additional equipment within the Premises during the Facility Term to the extent related to the general manufacturing purposes already authorized.

5. <u>Supply of Membranes</u>

- **Sale of Membrane**. Entegris agrees to sell to Millipore Flat Sheet UPE Membranes, in the amounts contained in Millipore's Releases, at all times during the Term and, in the event this Agreement is terminated by Entegris pursuant to Section 6.2 (iii), at all times following the Term through and including March 31, 2014 or until any event prior thereto that would have given rise to Entegris' right to terminate this Agreement pursuant to Section 6.2 (i) or (ii). Except as set forth in Section 5.4 hereof, neither party shall have any minimum or maximum purchase requirements for any or all of such Membranes hereunder, either per order or in the aggregate.
- **Terms of Sale.** Unless otherwise agreed by both parties in writing, this Agreement applies to all Releases placed by a Supplied Party with a Supplying Party during the Term. The terms and conditions of this Agreement shall apply to any Release, whether or not this Agreement or its terms and conditions are expressly referenced in the Release. All Membrane shall be tested, inspected and packaged for delivery by the Supplying Party as mutually agreed by the parties.
- **Priority.** Unless otherwise agreed by both parties in writing for a specific transaction, no inconsistent or additional term or condition in any Release, or in any acknowledgment, invoice or other document issued by a Supplying Party or its representative in connection with a particular purchase by a Supplied Party, shall be applicable to a transaction within the scope of this Agreement. Both

parties specifically agree that any terms and conditions in any such documents which are in any way inconsistent with this Agreement shall be inapplicable, and the terms of this Agreement shall govern.

5.4. Forecasts and Releases.

- Millipore will provide Entegris with a rolling one-year forecast of its demand for UPE Membranes hereunder, by calendar quarter (a "One-Year Forecast"), which will be updated on a quarterly basis, at least thirty (30) days prior to the start of each calendar quarter. The sub-forecast for the first three (3) months within any One-Year Forecast shall be referred to as a "3 Month Forecast". Millipore must provide Releases for delivery, during the three (3) months covered by any 3 Month Forecast, of UPE Membranes in at least those quantities set forth in such 3 Month Forecast. Except for such semi binding nature of the 3 Month Forecasts as described more specifically at 5.4.3 below, the One-Year Forecasts will be used for planning purposes only and are not binding. Entegris will ship UPE Membranes so as to arrive on the delivery date set forth in a Release, provided that the delivery date set forth in such Release is not less than thirty (30) days following the date Entegris receives such Release and provided that the quantities set forth in such Release, together with those in all other Releases calling for delivery during the same quarter, are not more than thirty percent (30%) greater than the quantities provided in the applicable 3 Month Forecast. For any Releases calling for quantities more than thirty percent (30%) greater than the quantities provided in such 3 Month Forecast, Entegris shall use commercially reasonable efforts to deliver such quantities within sixty (60) days following the date Entegris receives such release or as soon as practicable thereafter.
- 5.4.2 Within ten (10) days after receipt of each One Year Forecast, Entegris will provide a non-binding, good faith projection of its UPE Membrane manufacturing volume (broken down between philic and phobic Membranes), by calendar quarter, for the upcoming four calendar quarters. This information will be used by Millipore for budgeting of resources and revenue, and for the determination of budgeted Machine Hourly Rates for purposes of Section 7.1.2.
- 5.4.3 Millipore's requirements for membranes for an upcoming quarter shall be submitted to Entegris in the form of a firm forecast before the start of that quarter. Millipore shall be entitled to change the forecast without penalty or other change if the written cancellation notice is received by Entegris prior to the requested membrane lot being extruded. Millipore shall be entitled to increase volume of membrane requested during the quarter by issuing a forecast change in writing and Entegris agrees to make reasonable efforts to accommodate the requested additional volume. Actual purchase orders will be issued following lot completion in order to assure that ordered quantities match quantities available.

- **5.5. Prices; Delivery and Payment Terms.** Membrane prices shall be as set forth in Section 7.5 hereof. Payment terms for the sale of Membranes hereunder shall be as set forth in Section 7.6 hereof. Delivery terms for all Membranes will be FOB 80 Ashby Road, Bedford, MA.
- 5.6 <u>Continuing Supply Discussions</u>. The parties agree to commence good faith discussions in the first calendar half of 2012 regarding the possibility of continuing the supply of some Membranes from Entegris to Millipore beyond the Term of this Agreement on terms and conditions mutually acceptable to both parties, if any.

5.7 <u>Safety Stock</u>.

- **5.7.1** Each party acknowledges that the generation of a sufficient safety stock of UPE Membrane inventory meeting specifications and produced using a qualified process (the "Safety Stock") is appropriate. Accordingly, during the Term Millipore will determine routine safety stock levels required and include sufficient membrane quantities in quarterly forecasts to achieve the desired safety stock levels.
- **5.7.2** Commencing in the calendar quarter immediately following Millipore's receipt of the Move Notice from Entegris, Millipore will supplement each One-Year Forecast thereafter with an extended forecast of demand which Millipore reasonably believes it will require during the six month period immediately following the Move ("Move Buffer Stock"). Entegris will review such Move Buffer Stock forecast and communicate to Millipore any concerns Entegris may have regarding its ability to produce the requested quantity. The portion of the Move Buffer Stock that has been QA released as finished rollstock will be owned and carried by Millipore. If it is agreed that Entegris will hold some of the Move Buffer Stock as gel rolls, then Entegris will own and carry such inventory until such time as it is converted to acceptable finished rollstock.
- **5.7.3** Entegris will not commence the Move, including the reduction or cessation of production or the movement of any Entegris Equipment, until the parties have created an inventory of Safety Stock of at least the amount of the Move Buffer Stock. Until commencement of the Move, Entegris will make all commercially reasonable efforts to maintain the Move Buffer Stock at such amount.
- **5.7.4** The parties agree that production in any calendar quarter in excess of, or below, 200,000 feet shall increment or decrement, respectively, the amount of Safety Stock. In addition, material returns for verified quality issues related to the

portions of the process controlled by Entegris will reduce the amount of Safety Stock, foot for foot. The parties will review the amount of Safety Stock on a monthly basis and will mutually determine the then current level of Safety Stock. The parties acknowledge that the Safety Stock requirements described in this section are exclusively for the benefit of Millipore's supply needs. Millipore will provide the necessary storage space for any Safety Stock, including the Move Buffer Stock, which storage space shall be outside the then current Premises.

5.7.5 Although fully treated, finished Membrane is preferable, if the parties mutually agree that it is a necessity in order to facilitate the Move, up to fifteen percent (15%) of the Move Buffer Stock may be comprised of gel rolls, which are composed of Flat Sheet UPE Membrane that has not yet been processed through the Annealing Line. Any such gel rolls shall be processed at the Premises when such rolls are actually processed.

6. TERM AND TERMINATION

- **6.1.** The effective period of this Agreement (the "Term" or "Term of this Agreement") shall begin on the Effective Date and continue thereafter until March 31, 2014 or until earlier termination in accordance with Section 6.2. Any Release issued by a Supplied Party before the effective date of termination and in accordance with Section 5.4 hereof shall be fulfilled by the Supplying Party.
- **6.2.** Either party may terminate this Agreement prior to December 31, 2012 without prejudice to any rights or liabilities accruing up to the date of termination:
 - (i) in the event of a material breach by the other party of any of the terms and conditions of this Agreement, by giving the other party written notice of such breach, provided that such breach shall not have been cured within one hundred twenty (120) days following such notice; or,
 - (ii) immediately, by written notice thereof, if any of the following events or an event analogous thereto occurs:
 - a. an adjudication has been made that the other party is bankrupt or insolvent;
 - b. the other party has filed bankruptcy proceedings or has had such proceedings filed against it, except as part of a bona fide scheme for reorganization;
 - c. a receiver has been appointed for all or substantially all of the property of the other party;
 - d. the other party has assigned or attempted to assign this Agreement for the benefit of its creditors; or
 - e. the other party has begun any proceeding for the liquidation or winding up of its business affairs; or,

Page 15 of 50

- (iii) at any time for convenience of the terminating party upon twenty-four (24) months prior written notice to the other party.
- **6.3.** Termination under this Section 6 shall be in addition to and not a substitute for other rights or causes of action of the terminating party.
- **6.4.** Termination of this Agreement shall not in any way operate so as to impair or destroy any of the rights or remedies of either party, either at law or in equity, nor shall it relieve the parties of their obligations pursuant to Sections 1, 3.1, 4.5, 4.6, 4.7, 6, 8, 9, and 13 through 16 hereof, each of which shall survive the termination or expiration of this Agreement.

7. PRICES AND PAYMENTS

- **7.1. Machine Hourly Rates.** The hourly rates ("Machine Hourly Rates") for use by Entegris of Millipore's VMF4 Line, Film 1 Annealing Line and Mix Room equipment (the "Machines"), and the invoicing thereof, shall be as set forth in this Section 7.1. Machine usage (including production and research and development usage) shall be calculated in accordance with the formula set forth in **Annex 2**.
 - 7.1.1. Machine Hourly Rates. Commencing on the Effective Date, the Machine Hourly Rates set forth in Annex 1, subject to the annual adjustments as set forth below in this Section, shall be applicable from the Effective Date through the end of the period of the Term of this Agreement. The Machine Hourly Rates shall be adjusted annually as of January 1 of each year, beginning in calendar year 2010, to reflect (i) the then most recent August-to-following-August percentage changes, up or down, in the Producer Price Index for the industry group Pharmaceutical Preparation Manufacturing (series identification number PCU325412325412), (ii) any demonstrated increases in Millipore's costs associated directly with use of the Machines hereunder and which costs are extraordinary, not presently anticipated and not reflected in the Producer Price Index used pursuant to clause (i) above, and (iii) changes to depreciation charges as a result of capital improvements to the Machines for the production of Membranes, as set forth in Section 7.4.2; provided, however, that commencing on January 1, 2009 the Machine Hourly Rates for Millipore's VMF4 Line will not be subject to such annual changes and will remain at \$313.38 per hour through December 31, 2012 after which VMF4 Line hourly rates may be adjusted annually following the same approach used for other machine hourly rates as described above. Machine Hourly Rates to Entegris are independent of Millipore production volumes on the Machines. 7.1.2. <u>Invoicing for Machine Hourly Rates</u>. Millipore shall be entitled to invoice Entegris monthly for Machine usage following the end of each month. The Machine Hourly Rates used for the first two months of each calendar quarter shall be the average Machine Hourly

Page 16 of 50

Rates that would apply for the budgeted use of each Machine for such quarter, as calculated by Millipore based on the then most recent projections provided by Entegris pursuant to Section 5.4.2 hereof. The invoice for the third month of each calendar quarter shall be adjusted to effect a "true up" to actual Machine usage by invoicing for the net difference between the Machine Hourly Rate applicable to the actual hourly usage of the Machine in question for the entire quarter multiplied times the actual hours of usage of such Machine for the quarter less the amounts invoiced for usage of such Machine for the first two months of the quarter. This "true up" is so that Entegris effectively pays for all of a quarter's hours at the Machine Hourly Rates appropriate for those total hours.

- 7.2. Occupancy Rates. Commencing on the Effective Date, rates for occupancy and use of the Premises shall be as set forth in Exhibit B.
- **Support Rates.** Commencing on the Effective Date, rates for extraordinary Millipore supervisory, operational assistance and technical support (as described in Section 3.2 hereof) that are currently in use by the parties during 2005 shall be applicable from the Effective Date through December 31, 2008. Commencing on January 1, 2009, rates for extraordinary Millipore supervisory, operational assistance and technical support (as described in Section 3.2 hereof) shall be as shown in Annex 1. These rates will be adjusted annually as of January 1 of each year, beginning in calendar year 2010, to reflect (i) the then most recent August-to-following-August percentage changes, up or down, in the Producer Price Index for the industry group Pharmaceutical Preparation Manufacturing (series identification number PCU325412325412) and (ii) any demonstrated increases in Millipore's costs associated directly with providing such support hereunder and which costs are extraordinary, not presently anticipated and not reflected in the Producer Price Index used pursuant to clause (i) above.

7.4. <u>Impact of Capital Investment on Prices.</u>

7.4.1. Entegris shall be responsible for funding required capital improvements to the Entegris Equipment for production of hydrophobic Membranes. Millipore shall be responsible for funding required capital improvements to the Millipore Equipment for production of hydrophilic Membranes. To the extent that capital improvements are required for the Millipore Equipment used in the production of hydrophobic Membranes, responsibility for funding that capital improvement shall be mutually agreed upon in accordance with Section I.1.7 of the Supplement. If one party funds the purchase of additional equipment for use in manufacture of UPE Membranes at the Premises, that equipment shall be owned by the funding party.

Page 17 of 50

- **7.4.2.** Changes to depreciation charges as a result of capital improvements to the Equipment for the production of Membranes will be reflected in the appropriate work center rates and would be part of the rate adjustment process set forth in Sections 7.1.2 and 7.3. Millipore and Entegris each agree to inform the other of capital purchases that may eventually affect proposed rate/price changes hereunder no later than the time that purchase orders for such capital purchases are placed.
- **7.4.3.** Millipore shall have approval authority with respect to any Entegris proposed modifications to VMF 4 equipment for UPE Membrane processing. Millipore agrees that proposed modifications to the VMF4 equipment for non-Entegris UPE Membrane processing that could influence the processing of UPE Membranes on the VMF 4 equipment shall be reviewed with Entegris and shall be subject to mutual agreement.
- **7.4.4.** Millipore shall have approval authority for any Entegris proposed modifications to Film 1 manufacturing process for UPE Membranes. Financial impact of those modifications would be mutually agreed upon prior to proceeding.
- **7.4.5.** In the event of a process change to either the Film 1 or VMF4 lines, Millipore shall be responsible for change control management for any impact on Millipore products. Likewise, Entegris would be responsible for change control management for any impact on Entegris products.

7.5. <u>Membrane Prices</u>.

- **7.5.1.** Intentionally Omitted.
- **7.5.2.** Commencing on the Effective Date through December 31, 2009, until adjusted pursuant to Section 7.5.3, prices for Flat Sheet UPE Membranes to Millipore produced at the Premises, together with up to eight (8) rolls produced at the New Entegris Facility that are used to ascertain basic feasibility, shall be at Entegris' Manufacturing cost plus fifteen percent (15%). For Flat Sheet UPE Membranes produced at the New Entegris Facility (other than the rolls for feasibility evaluation as described and priced above), prices shall as set forth on Annex 3 for the Term of this Agreement. The parties acknowledge that all Flat Sheet UPE Membranes will be produced at the New Entegris Facility after June 30, 2012.
- **7.5.3.** Entegris has notified Millipore as to the adjusted Manufacturing Cost to be in effect for calendar year 2009 for the Membranes to be supplied by Entegris that will be produced at the Premises and shall make available an "open-book" review of such Manufacturing Cost. Membrane Prices hereunder will be adjusted as of January 1, 2010 to reflect such adjusted Manufacturing Cost. Without limiting the factors involved in determining Manufacturing Cost, the Manufacturing Cost of UPE Membranes shall take into account the budgeted Machine Hourly Rates based on projected annual UPE Membrane manufacturing volume.

7.6. Payment of Invoices. All amounts payable by either party to the other pursuant to this Agreement, except for amounts payable as Rent, shall be payable within forty-five (45) days following the later of (i) receipt of ordered Membranes, performance of services or Machine usage, and (ii) receipt of invoice. All payments shall be made in U.S. Dollars. Any late payments shall be subject to interest at a rate of twelve percent (12%) per annum.

8. NON-COMPETITION

Except as otherwise provided in Article 5 hereof or elsewhere in this Agreement:

- 8.1. Millipore agrees that neither it nor any of its Affiliated Companies will (i) sell outside of the Millipore Core Business any UPE Membranes or UPE Products, or (ii) sell any UPE Membranes or UPE Products to any distributor, OEM manufacturer or other third party that has rights to, or that Millipore or any such Affiliated Company has reason to believe will, resell such UPE Membranes or UPE Products outside of the Millipore Core Business or sell other products which include UPE Membranes or UPE Products as materials or components outside of the Millipore Core Business.
- 8.2. Entegris agrees that neither it nor any of its Affiliated Companies will (i) sell into the Millipore Core Business any Other UPE Membranes or Other UPE Products, or (ii) sell any Other UPE Membranes or Other UPE Products to any distributor, OEM manufacturer or other third party that has rights to, or that Entegris or any such Affiliated Company has reason to believe will, resell such Other UPE Membranes or Other UPE Products into the Millipore Core Business or sell other products which include Other UPE Membranes or Other UPE Products as materials or components into the Millipore Core Business.
- **8.3.** Intentionally Omitted.
- **8.4.** Intentionally Omitted.
- 8.5. Entegris agrees that neither it nor any of its Affiliated Companies will (i) sell outside of the Entegris Core Business any Treated Entegris Membranes or Treated Entegris Products, or (ii) sell any Treated Entegris Membranes or Treated Entegris Products to any distributor, OEM manufacturer or other third party that has rights to, or that Entegris or any such Affiliated Company has reason to believe will, resell such Treated Entegris Membranes or Treated Entegris Products outside of the Entegris Core Business or sell other products which include Treated Entegris Membranes or Treated Entegris Products as materials or components outside of the Entegris Core Business.

Page 19 of 50

- 8.6. In the event that either party discovers any distribution arrangements pre-existing the Original Contract Date that would conflict with the provisions of this Agreement, the parties agree that any such pre-existing arrangements shall not constitute a breach hereunder, and they further agree: (i) to use reasonable commercial efforts to cause any such terms of distribution agreements that are inconsistent with the provisions contained herein to be amended so as to be consistent with these provisions, (ii) not to amend any distribution agreements following the date of this Agreement so as to be inconsistent with such provisions, and (iii) not to renew or enter into any distribution agreements or other agreements containing terms inconsistent with the provisions contained herein following the date of this Agreement.
- 8.7. It is acknowledged and accepted that either party or its Affiliated Companies may from time to time hereafter unintentionally make sales that would be prohibited in accordance with Sections 8.1 through 8.5 ("Sales Outside Field"). Accordingly, notwithstanding Sections 8.1 through 8.5, each party agrees not to actively market or attempt to make Sales Outside Field, *provided* that for any Sales Outside Field that are nevertheless made by and known to a party, such party shall, within forty-five (45) days following the end of each calendar year in which such Sales Outside Field were made, provide an accounting of its Gross Margins on such Sales Outside Field during such calendar year (such accounting to include the total amount of such Sales Outside Field, and detail regarding the customers to which such Sales Outside Field were made) and payment of the amount of such Gross Margins to the other party. Other than such accountings, neither party shall be liable for any commission, payment, remittance, accrual or obligation or incur any other liability to the other party with respect to any such Sales Outside Field. For purposes of this Section, "Gross Margins" on a party's Sales Outside Field shall mean the sale price, net of discounts and other sales deductions, of a Membrane or product sold Outside Field, less such party's fully burdened manufacturing cost of such Membrane or product (which for Membranes purchased from the other party hereunder shall mean the price paid to the other party for such Membranes).

9. WARRANTIES AND INDEMNIFICATION

- **9.1.** Each Supplying Party warrants to the corresponding Supplied Party that:
 - (i) All Membranes supplied to the Supplied Party hereunder shall conform to the specifications for such Membranes as in effect as of the date of this Agreement and as provided to the Supplied Party, as such specifications may be amended as agreed by the parties;
 - (ii) All Membranes supplied hereunder shall be free of defects in materials and workmanship; and

Page 20 of 50

- (iii) It will abide by all applicable laws and regulations in manufacturing and supplying Membranes pursuant to this Agreement.
- **9.2.** In the event of a breach of the foregoing warranties, the Supplying Party's sole obligation to the Supplied Party shall be to repair, replace or refund, at the Supplying Party's option, any non-conforming Membranes.
- 9.3. THE SUPPLYING PARTY MAKES NO OTHER WARRANTY, EXPRESSED OR IMPLIED, INCLUDING WITHOUT LIMITATION ANY WARRANTY OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR USE. FURTHERMORE, THE SUPPLYING PARTY SHALL NOT BE LIABLE FOR CONSEQUENTIAL, INCIDENTAL, SPECIAL OR ANY OTHER INDIRECT DAMAGES RESULTING FROM ECONOMIC LOSS OR PROPERTY DAMAGE SUSTAINED BY THE SUPPLIED PARTY FROM THE USE OF THE SUPPLIED MEMBRANES.
- **9.4.** Each Supplying Party agrees to indemnify and hold the corresponding Supplied Party harmless from and against any claim or legal action by a third party against such Supplied Party (including reasonable attorneys' fees associated therewith) based on damages incurred as a result of property damages, personal injury or death, to the proportionate extent arising from a breach of any of the above warranties of the Supplying Party or from the Supplying Party's negligent action or omission.
- 9.5. Without limiting any other rights or remedies that a Supplied Party may have, if such Supplied Party determines that delivered Membranes do not conform to the agreed specifications for such Membranes, then such Supplied Party may reject or withdraw its acceptance thereof and shall notify the Supplying Party in writing of such nonconformity or error within thirty (30) days from receipt of such Membranes by the Supplied Party. The Supplied Party may subject any Membrane to internal testing for purposes of determining conformity to specifications. The Supplying Party shall have fifteen (15) days after receipt of written notice of nonconformity or error to replace nonconforming Membranes at the expense of the Supplying Party. If so directed by the Supplying Party, the Supplied Party shall return nonconforming Membranes to the Supplying Party's manufacturing facilities, at the Supplying Party's expense and using such carrier and such delivery dates and terms as the Supplying Party shall reasonably specify.
- **9.6.** The parties agree to have their representatives meet at least once every three (3) months (unless otherwise agreed) to review compliance with the manufacturing, specifications, product quality, forecasting and delivery terms set forth in this Agreement, and to agree on any necessary corrective actions or modifications to the Supplement as then in effect.

10. <u>MEMBRANE MODIFICATIONS; NEW MEMBRANES.</u>

- **10.1.** Each Supplying Party agrees that it will not substantially change the Membranes that it will supply hereunder or their formulation, manufacturing or testing processes, process equipment, other aspects of form, fit or function, or production location, unless the Supplied Party approves such change in writing, which approval may require formal validation and qualification and possibly customer notification. The implementation of any such accepted changes shall be subject to the parties' agreement on any change in price or other terms of supply as may be necessitated or requested by a party as a result of such change.
- 10.2. If any new or improved UPE Membranes result from research and development work that may be conducted by either Millipore or Entegris during the Term of this Agreement, or are requested by Millipore to be added to the supply provisions hereof, and are agreed to by Entegris, and are technically feasible for Entegris to manufacture, it is intended that such UPE Membranes be added to this Agreement both in terms of Entegris' supply to Millipore and Entegris' manufacture of such UPE Membranes, and they shall be so added to this Agreement upon agreement by the parties as to specifications and pricing, which pricing shall be consistent in methodology with the pricing hereunder.

11. ACCESS TO FACILITIES

At any time during the Term, upon reasonable advance notice by a Supplied Party, such Supplied Party's authorized representatives and customers (subject to appropriate confidentiality obligations) shall be provided access to the facilities of the Supplying Party to audit or verify conformity with applicable laws and regulations and mutually agreed to quality standards. During the Facility Term of this Agreement, Millipore's authorized representatives shall be provided access to the Premises for the purpose of auditing or troubleshooting (to be coordinated with Entegris) of technical problems with UPE Membranes or their manufacture. Also, Millipore and Entegris customers (subject to appropriate confidentiality obligations and on reasonable advance notice, and for the purposes indicated above) shall be provided reasonable access, respectively, to the Premises (during the Facility Term) and to the New Entegris Facility and to the areas of Millipore's 80 Ashby Road facility where the Millipore Equipment is located and used (limited to the VMF4 Line after the Facility Term) up to the Term of this Agreement.

12. INSURANCE RELATED TO MEMBRANES

Each Supplying Party agrees to procure and maintain, at all times during the Term, product liability insurance with respect to the Membranes supplied by it (Broad Form Vendor's Endorsement) and contractual liability coverage, with the minimum limits of \$5,000,000 (Five Million Dollars). Each Supplying Party shall, upon request by the Supplied Party, furnish to the Supplied Party a certificate of insurance evidencing the foregoing coverage and limits. The insurance provider shall not be changed without providing the Supplied Party with ten (10) days' prior written notice.

13. <u>CONFIDENTIALITY</u>

- 13.1. Confidential Information. For the purpose of this Agreement the term "Confidential Information" means Information which is not otherwise in the public domain and of which the owner actively undertakes to restrict or control the disclosure to persons or entities other than Millipore or Entegris or their Subsidiaries in a manner reasonably intended to maintain its confidentiality, and which: (i) the party owning or disclosing Confidential Information ("Disclosing Party") disclosed to the non-owning party or recipient of the Confidential Information ("Receiving Party") or the Receiving Party had access to on or before the Original Contract Date; (ii) is contained in or referred to by this Agreement or any exhibit or annex hereto and is known to or in the possession of the Receiving Party as of the Effective Date; or (iii) is disclosed to the Receiving Party pursuant to this Agreement during the Term (the "Disclosure Period"). Confidential Information may include information relating to, by way of example, research, products, services, customers, markets, software, developments, inventions, manufacturing processes, designs, drawings, engineering, marketing or finances, and may be in writing, disclosed orally or learned by inspection of computer programming code, equipment or facilities. Confidential Information of third parties that is known to, in the possession of or acquired by a Receiving Party pursuant to a relationship with the Disclosing Party shall be deemed to be the Disclosing Party's Confidential Information for purposes of this Section 13.
 - 13.1.1. <u>Highly Confidential Information</u> means Confidential Information that is technical know-how and trade secrets relating to:

 (i) Information relating to manufacturing processes or procedures with respect to devices or other products that are commercially released or for which substantial steps have been taken towards commercialization as of the Effective Date; (ii) Information generated by research and development activities; (iii) chemical and other scientific formulae used for the manufacture or treatment of membranes or other separations media or of devices or other products that are commercially released or for which substantial steps have been taken towards commercialization as of the Effective Date; or (iv) any other Information which Millipore and Entegris agree is Highly Confidential hereunder.
 - **13.1.2.** Exclusions from Confidential Information. Notwithstanding the foregoing provisions of this Section 13.1, Confidential Information shall exclude information that: (i) was in the Receiving Party's possession before receipt from the Disclosing Party and obtained from a source other than the Disclosing Party and other than through the prior relationship of the Disclosing Party and the Receiving Party

before the Original Contract Date; (ii) is or becomes a matter of public knowledge through no fault of the Receiving Party; (iii) is rightfully received by the Receiving Party from a third party without a duty of confidentiality; (iv) is disclosed by the Disclosing Party to a third party without a duty of confidentiality on such third party; (v) is independently developed by the Receiving Party; or (vi) is publicly disclosed by the Receiving Party with the Disclosing Party's prior written approval.

- 13.2. <u>Confidentiality And Non-Use Obligations</u>. During the Confidentiality Period (as defined in Section 13.3 below), the Receiving Party shall (i) protect the Confidential Information of the Disclosing Party by using the same degree of care, but no less than a reasonable degree of care, to prevent the unauthorized use, dissemination, or publication of the Confidential Information as Receiving Party uses to protect its own confidential information of a like nature, (ii) not use such Confidential Information in violation of any use restriction herein, and (iii) not disclose such Confidential Information to any third party, except as expressly permitted under this Agreement or in any other agreements entered into between the parties in writing, without prior written consent of the Disclosing Party.
- **Duration of Confidentiality Obligations.** The confidentiality obligations provided for in this Section 13 shall continue in effect for the following periods (the "Confidentiality Period"): (i) with respect to Confidential Information that is not Highly Confidential Information, for a period of five (5) years following either (A) the Effective Date with respect to Confidential Information of the Disclosing Party that is known to or in the possession of the Receiving Party as of the Effective Date or (B) the date of disclosure with respect to Confidential Information that was or will be disclosed by the Disclosing Party to the Receiving Party after the Effective Date but before the expiration of the Disclosure Period (as defined in Subsection 13.1 above); and (ii) with respect to Highly Confidential Information, in perpetuity. The obligations set forth in this Section 13 shall survive any termination of this Agreement.
- 13.4. Compelled Disclosure. If the Receiving Party or any of its respective Subsidiaries believes that it (i) is legally obligated to disclose, or (ii) will be compelled by a court or other authority of competent jurisdiction to disclose, Confidential Information of the Disclosing Party, it shall give the Disclosing Party prompt written notice so that the Disclosing Party may take steps to oppose such disclosure and cooperate with the Disclosing Party in its attempts to oppose such disclosure. If the Receiving Party complies with the preceding sentence, it shall not be prohibited from complying with such requirement to disclose, but shall take all reasonable steps to make such disclosure subject to a suitable protective order or otherwise to prevent unrestricted or public disclosure.

- **13.5.** *No Restriction on Disclosing Party.* Nothing in this Section 13 shall restrict the Disclosing Party from using, disclosing, or disseminating its own Confidential Information in any way.
- 13.6. <u>Disclaimer of Warranties as to Confidential Information</u>. EACH PARTY ACKNOWLEDGES AND AGREES THAT ALL CONFIDENTIAL INFORMATION IS PROVIDED ON AN "AS IS, WHERE IS" BASIS AND THAT NEITHER PARTY NOR ANY OF ITS SUBSIDIARIES HAS MADE OR WILL MAKE ANY WARRANTY WHATSOEVER, EXPRESS, IMPLIED OR STATUTORY, INCLUDING, WITHOUT LIMITATION, ANY IMPLIED WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, TITLE, ENFORCEABILITY OR NON-INFRINGEMENT.

14. NO OTHER RIGHTS

Neither party shall have any rights hereunder to any patents or other intellectual property of the other party, except as specifically set forth herein. Without limiting the generality of the foregoing, neither party shall have any rights to use any trademarks of the other party for any purpose in connection with the Membranes to be manufactured or supplied hereunder.

15. DISPUTE RESOLUTION

- **Discussion of Parties.** In the event of a dispute between the parties arising out of or related to this Agreement (the "Dispute"), a party seeking to resolve the Dispute shall give written notice to the other party, describing briefly the nature of the Dispute and its claim and identifying an individual with authority to settle the Dispute on its behalf. The party receiving such notice shall have five (5) business days within which to designate, in a written notice given to the initiating party, an individual with authority to settle the Dispute on its behalf. Neither of such authorized individuals shall have had direct substantive involvement in the matters involved in the Dispute. The authorized individuals shall make such investigation as they deem appropriate and thereafter promptly (but in no event later than thirty (30) days from the date of the initiating party's notice) shall commence discussions concerning resolution of the Dispute.
- 15.2. If the Dispute has not been resolved within thirty (30) days from the commencement of discussions, it shall be submitted to final and binding arbitration under the then current Commercial Arbitration Rules of the American Arbitration Association ("AAA"), by one (1) arbitrator in Boston, Massachusetts. Such arbitrator shall be selected by the mutual agreement of the parties or, failing such agreement, shall be selected according to the aforesaid AAA rules. The arbitrator will be instructed to prepare and deliver a written, reasoned opinion stating its decision within thirty (30) days of the completion of the arbitration. The prevailing party in such arbitration shall be entitled to expenses, including costs and reasonable attorneys' and other professional fees,

incurred in connection with the arbitration. The decision of the arbitrator shall be final and non-appealable and may be enforced in any court of competent jurisdiction.

Continuity of Service and Performance. Unless otherwise agreed in writing, the parties will continue to provide service and honor all other commitments under this Agreement during the course of dispute resolution pursuant to the provisions of this Section 15 with respect to all matters not subject to such dispute, controversy or claim.

16. GENERAL PROVISIONS

16.1. Notices. Any notice or other communication required or permitted to be given by either party pursuant to the terms of this Agreement shall be in writing and shall be deemed given if and when delivered by hand or sent by certified mail, return receipt requested, overnight courier, confirmed telecopy, or confirmed electronic mail transmission, addressed as follows:

If to Millipore: Millipore Corporation

290 Concord Road Billerica, MA 01821

Attn: Vice President, Global Operations

Fax: (978) 715-1385

with a copy to: Millipore Corporation

290 Concord Road Billerica, MA 01821 Attn: General Counsel Fax: (978) 715-1382

If to Entegris: Entegris, Inc.

129 Concord Road Billerica, MA 01821 Attn: Ex. VP & COO Fax: (978) 436-6739

with a copy to: Entegris, Inc.

129 Concord Road Billerica, MA 01821 Attn: General Counsel Fax: (978) 436-6739

or to such electronic mail address as may be specified by an addressee party to the other party by one of the other means provided above, or to such other address, telecopy number or electronic mail address as may be specified by an addressee party to the other by one of the means provided above.

- **Force Majeure.** The obligations of a party under this Agreement will be suspended to the extent that it is wholly or partially precluded from complying with its obligations under this Agreement by force majeure. Force majeure includes, but is not restricted to, fire, storm, flood, earthquake, explosion, accident, act of the public enemy, war, rebellion, insurrection, sabotage, epidemic, quarantine restriction, labor dispute, labor shortage, transportation embargo or failure or delay in transportation, act of God, act (including laws, regulations, disapprovals or failure to approve) of any government agency, whether national, municipal, or otherwise. During the existence of any such force majeure condition, the affected party shall nevertheless use its best efforts to remove the cause thereof.
- **Entire Agreement; Old Agreement.** This Agreement, including Exhibits A, B, and C and Annexes 1 and 2, attached hereto, is the entire agreement between the parties with respect to the subject matter hereof, and supersedes any prior negotiations and agreements or understandings and any contemporaneous oral agreements or understandings with respect to the subject matter hereof. Without limiting the generality of the foregoing, the Old Agreement shall be deemed replaced in its entirety by this Agreement as of the Effective Date, provided that the Old Agreement shall continue to apply to all activities or events that occurred prior to the Effective Date.
- **Governing Law.** This Agreement shall be construed in accordance with and all Disputes hereunder shall be governed by the laws of the Commonwealth of Massachusetts as applied to transactions taking place wholly within Massachusetts between Massachusetts residents. The Superior Court of Middlesex County and/or the United States District Court for the District of Massachusetts shall have jurisdiction and venue over all Disputes between the parties that are permitted to be brought in a court of law pursuant to Section 15 above.
- **16.5.** *Counterparts*. This Agreement and the Exhibits and Annexes hereto and the other documents referred to herein, may be executed in counterparts, each of which shall be deemed to be an original but all of which shall constitute one and the same agreement.
- **Binding Effect; Assignment**. This Agreement shall inure to the benefit of and be binding upon the parties hereto and their respective legal representatives and successors, and nothing in this Agreement, express or implied, is intended to confer upon any other person any rights or remedies of any nature whatsoever under or by reason of this Agreement. Neither party may assign this Agreement or any rights or obligations hereunder, without the prior written consent of the other party, and any such assignment shall be void. Notwithstanding the foregoing, either party may assign this Agreement and all (but not less than all) of its rights and obligations hereunder to a purchaser or transferee of, or other successor to, substantially all of its business.

- **Severability.** If any term or other provision of this Agreement or the Exhibits or Annexes attached hereto is determined by a court, administrative agency or arbitrator to be invalid, illegal or incapable of being enforced by any rule of law or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to either party. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the parties hereto shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner to the end that transactions contemplated hereby are fulfilled to the fullest extent possible.
- **Failure or Indulgence Not Waiver; Remedies Cumulative.** No failure or delay on the part of either party hereto in the exercise of any right hereunder shall impair such right or be construed to be a waiver of, or acquiescence in, any breach of any representation, warranty or agreement herein, nor shall any single or partial exercise of any such right preclude other or further exercise thereof or of any other right. All rights and remedies existing under this Agreement or the Exhibits and Annexes attached hereto are cumulative to, and not exclusive of, any rights or remedies otherwise available.
- **16.9. Amendment**. No change or amendment will be made to this Agreement or the Exhibits or Annexes attached hereto except by an instrument in writing signed on behalf of each of the parties to such agreement.
- **Authority.** Each of the parties hereto represents to the other that **(a)** it has the corporate or other requisite power and authority to execute, deliver and perform this Agreement, **(b)** the execution, delivery and performance of this Agreement by it have been duly authorized by all necessary corporate or other actions, **(c)** it has duly and validly executed and delivered this Agreement, and **(d)** this Agreement is a legal, valid and binding obligation, enforceable against it in accordance with its terms subject to applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting creditors' rights generally and general equity principles.
- **16.11.** *Interpretation.* The headings contained in this Agreement or in any Exhibit hereto are for reference purposes only and shall not be conclusive as to the meaning or interpretation of this Agreement. When a reference is made in this Agreement to a Section, Exhibit or Annex, such reference shall be to a Section, Exhibit or Annex of this Agreement unless otherwise indicated.

16.12. *Exhibits and Annexes*. This Agreement includes the following Exhibits and Annexes, each of which constitutes an integral component part of this Agreement:

Exhibit A. Membranes Covered by Agreement
Exhibit B Lease Terms including Attachment A

Exhibit C The Supplement

Annex 1 Machine Hourly Rates and Support Rates
Annex 2 Formula for Calculating Machine Usage Hours
Annex 3 Membrane Prices from New Entegris Facility

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed by their duly authorized representatives effective as of the Effective Date.

MILLIPORE CORPORATION

ENTEGRIS, INC.

/s/ Peter C. Kershaw. By: /s/ Bertrand Loy.

Name: Peter C. Kershaw Name: Bertrand Loy

Title: Corporate Vice President, Global Operations Title: Executive Vice President and Chief Operating Officer

Page 29 of 50

Exhibit A Membranes Covered By Agreement

Membrane	Description	Note
	Other Flat Sheet UPE Membranes	
UPBP00000	1.0 um phobic, thick UPE	
DOHP00000	0.65 um phobic, thick UPE	
UPDP	0.65 um phobic, thick UPE	With Interleaf
UPHP00000	0.45 um phobic, thick UPE	With Interleaf
UPHP000HC	0.45 um phobic, thick UPE	With Interleaf
UPHP000NI	0.45 um phobic, thick UPE	
UPGP00000	0.2 um phobic, thick UPE	
UPGP000HC	0.2 um phobic, thick UPE	With Interleaf
UPVP00000	0.1 um phobic, thick UPE	
UPVP00000HF	0.1 um phobic, thick UPE	
UPZP00000	0.05 um phobic, thick UPE	
UPZP000HC	0.05 um phobic, thick UPE	With Interleaf
GOHT	0.25 um phobic, thin UPE	
TPGP00000	0.2 um phobic, thin UPE	
VOHT	0.15 um phobic, thin UPE	
TPVP00000	0.1 um phobic, thin UPE	
ZOHT	0.05 um phobic, thin UPE	
TPZP00000	0.05 um phobic, thin UPE	
TPLP	0.04 um phobic, thin UPE	
TPYP00000	0.03 um phobic, thin UPE	
TPXP00000	0.02 um phobic, thin UPE	
TPTP00000	0.01 um phobic, thin UPE	
	Treated Flat Sheet UPE Membranes	
JOTD00000	0.25 um Charged, philic UPE	
WOTD00000	0.15 um Charged, philic UPE	
ZOTD00000	0.05 um Charged, philic UPE	
WOTN00000	0.15 um Charged, neutral UPE	
VOTN00000	0.10 um Charged, neutral UPE	
ZOTN00000	0.05 um Charged, neutral UPE	
LOTN00000	0.04 um Charged, neutral UPE	
JETA00000	0.25 um anionic UPE	
WETA00000	0.15 um anionic UPE	
ZETA00000	0.05 um anionic UPE	
LETA00000	0.04 um anionic UPE	
ZOTAM0000	0.05 um PCM modified UPE	
ZOTN00000	PCM modified UPE	
YOTAM0000	0.02 um PCM modified UPE	
1 3 17 11110000	o.o. am i om modifica of L	

EXHIBIT B

Lease Terms

- 1. Millipore hereby leases to Entegris, and Entegris hereby leases from Millipore, the areas within Buildings "C", "D" and "F" of Millipore's facility at 80 Ashby Road, Bedford, MA, as more specifically described and shown in Attachment A to the terms set forth in this Exhibit B (collectively, the "Lease Terms"), as such areas may be added to, reduced or substituted for as provided in these Lease Terms (the "Premises"), including as specifically set forth in Section 3.3, solely for purposes of its manufacture of UPE Membranes, and PFA Hollow Fiber Membranes and warehousing of raw materials and components used in such manufacture, and for no other purposes except as Millipore may specifically authorize in its sole discretion.
- 2. The Lease of the Premises shall commence as of the Effective Date and continue through the end of the Facility Term, or until such time as Entegris has removed all Entegris Equipment from the Premises in accordance with the terms of Section 4.6.1 of this Agreement. Subject to achievement of the required Safety Stock as set forth in the Agreement at section 5.7, Entegris shall have vacated the Premises by the end of the Facility Term.
- 3. Entegris shall pay to Millipore rent/occupancy charges for the Premises ("Rent") as follows:
- From the Effective Date through December 31, 2009, Rent shall be \$ 91.14 per square foot of the Premises. Rent shall be adjusted annually thereafter, on a square-footage basis, on January 1 of each year, beginning with 2010, as follows: (a) The utility portion of the Rent (which for calculation of the current Rent shall be 39.8% of the total Rent) will be adjusted by the same percentage as the then most recent August-to-following-August percentage change in the average actual cost per kilowatt hour paid by Millipore for the 80 Ashby Road, Bedford facility (e.g. for calendar year 2010, the percentage change in such average cost from August 2008 to August 2009); (b) the remainder of the Rent (which for calculation of the current Rent shall be 60.2% of the total Rent) will be adjusted by the same percentage change as the then most recent August-to-following-August percentage change in the Producer Price Index for the industry group Pharmaceutical Preparation Manufacturing (series identification number PCU325412325412) (e.g. for calendar year 2010, the percentage change in such index from August 2008 to August 2009). Rent shall be payable by Entegris monthly on or before the fifth day of each month.
- 3.2 It is acknowledged and agreed that in the event that the square footage of the Premises is adjusted during the Facility Term as a result of a shift of production as provided in this Agreement or pursuant to Section 3.3, the corresponding Rent due to Millipore hereunder shall be adjusted accordingly.

Page 31 of 50

- The parties acknowledge that implementation of the Move will likely result in decreasing production by Entegris at the Premises during the Facility Term and therefore commensurate reductions in Entegris' need for the full original Premises. Accordingly, at least ten (10) days prior to the end of any month during which Entegris believes it has vacated any portion of the Premises, Entegris will notify Millipore of such occurrence in writing, such notice to include with reasonable specificity the portion of the Premises vacated and the square footage proposed to be surrendered. Upon receipt of the notice, Millipore will promptly review in good faith with Entegris the vacated portion of the Premises to confirm that such space has been vacated and, to the extent applicable, complies with Section 7.6 with respect to the condition of the portion of vacated Premises. Upon the parties' mutual agreement of the actual surrender and the square footage affected, the parties will agree on the revised square footage of the Premises and amend Attachment A to reflect such reduce square footage, to be effective for the ensuing month for purposes of calculating Rent pursuant to Sections 3.1 and 3.2 thereafter until any further reductions occur.
- At all times during the Facility Term, those employees or consultants of Entegris that are involved regularly or from time to time in the manufacture of 4. UPE Membranes and reasonably need access to the Premises for such purposes, and that are not objectionable to Millipore in its sole reasonable discretion, ("Entegris Permitted Persons") shall have access to the Premises for the purposes of manufacturing of UPE Membranes and Treated Other Entegris Membranes and warehousing of raw materials and components used in such manufacture, including reasonable ingress to and egress from the Premises through other parts of Millipore's facility at 80 Ashby Road. Notwithstanding, to the extent that Millipore believes in its sole reasonable discretion that an individual or individuals are objectionable and hence prohibited from entering the Premises, it shall provide Entegris a written summary of the reasons for, and basis of, such belief within three (3) days of notifying Entegris of such a determination. Without limiting the generality of the foregoing, Millipore shall provide security badges to such Entegris Permitted Persons enabling their entry to the Premises. Entegris agrees to provide an initial list of proposed Entegris Permitted Persons to Millipore's facilities personnel (as shall be identified to Entegris by Millipore) promptly after the execution of this Agreement, and to promptly update such list from time to time as necessary with any additions or deletions. In addition, Entegris Permitted Persons shall be, to the same extent, on the same terms, and subject to the same rules and restrictions, as provided to Millipore employees in general, (i) provided access to and use of restrooms, parking areas, driveways and footways, other common areas reasonably necessary to be used by such Entegris Permitted Persons, and conference rooms as reasonably necessary for meetings relating to UPE Membranes manufacture, and (ii) permitted reasonable use of mailroom services, telephone and voicemail systems, information services, fax, copy machines, cafeteria/refreshment services, and the like. Without limiting the generality of the parties' respective obligations of confidentiality and non-

described in this Agreement, Entegris hereby agrees that any Confidential Information of Millipore that the Entegris Permitted Persons gain access to as a result of their presence in Millipore's facility shall be subject to strict obligations of confidentiality and non-use as provided in, and subject to the limitations and other provisions of, Section 13 of this Agreement.

- 5. Millipore shall have the right, upon at least twenty-four (24) months written notice to Entegris, to relocate all or any portion of the Premises to any other location of comparable size and utility within its 80 Ashby Road facility. The parties shall cooperate in enabling any such relocation to occur in a coordinated manner so as to minimize interruption both to Entegris' manufacturing activities hereunder and Millipore manufacturing and other activities at such facility. Millipore shall reimburse Entegris for all reasonable and substantiated costs incurred by Entegris in connection with any such relocation. The definition of the "Premises" shall be modified to include any such relocated space.
- 6. Entegris shall comply with all federal, state or local statutes, ordinances, codes, rules, regulations, orders and decrees ("Laws") applicable to the conduct of its business on the Premises. Entegris shall not cause or permit any hazardous substances to be transported to or from, brought upon, kept, stored or used in or about the Premises by, its agents, employees, contractors or invitees, except for such hazardous substances as are reasonably necessary in the performance of the activities permitted under this Agreement. Any hazardous substance permitted pursuant to the foregoing sentence and all containers therefore shall be used, kept, stored, transported and disposed of in a manner that strictly complies with all applicable Laws relating thereto as now or at any time hereafter during the Facility Term may be in effect. Except as may be otherwise agreed between the parties in writing, during the Facility Term responsibility for regulatory, health, safety and environmental compliance, as well as hazardous and solid waste disposal, generated by Entegris' use of the Premises and Entegris' manufacture of UPE Membranes and Treated Other Entegris Membranes shall be allocated in accordance with Section I.4 of the Supplement.
- 7. Repairs and Alterations; Condition of Premises
- 7.1 Entegris agrees that it shall, throughout the Facility Term, maintain the Premises in good and operable repair and in such condition as may be required by any laws, ordinances, regulations or requirements of any public authorities having jurisdiction and shall not permit or allow to remain any waste or damage to any portion of the Premises, except as may be permitted herein, provided, however, that Millipore shall be responsible for maintaining, keeping in good repair and, if necessary, replacing the foundation, the exterior and interior walls, the roof, floors and ceilings, the heating, ventilating and air-conditioning system, all utility systems (including water, gas, electricity, drainage, and sewerage), conduits, fixtures and equipment, all meters and all other fixtures and equipment within or appurtenant to the Premises; provided, further, that Entegris shall promptly

reimburse Millipore, as additional Rent, for the cost of any such maintenance, repair or replacement to the extent necessitated or occasioned by any omission, negligence, fault or act of Entegris, its partners, officers, employees, agents, independent contractors or invitees; provided, however, that any maintenance, repair or replacement necessitated by fire or other casualty shall be subject to the provisions of Section 10 below. Entegris' maintenance of the Premises shall include, without limitation, cleaning the Premises regularly. If Entegris fails to perform such maintenance, within fifteen (15) days after written notice from Millipore (or as promptly as practicable if Entegris' failure to make such repairs or replacements or perform such maintenance constitutes an emergency or poses a threat to persons or property), then Millipore may elect to make the same at Entegris' cost (without creating any obligation on behalf of Millipore to do so). If Millipore fails to perform, or commence performance of, any maintenance, replacement or repair for which it is obligated under these Lease Terms within fifteen (15) days after written notice from Entegris (or as promptly as practicable if Millipore's failure to make such repairs or replacements or perform such maintenance constitutes an emergency or poses a threat to persons or property), then Entegris may elect to make the same at Millipore's cost (without creating any obligation on behalf of Entegris to do so).

- 7.2 Entegris agrees that it will not make any installations, alterations, additions or improvements to or upon the Premises (including without limitation the installation of signs) without the prior written consent of Millipore, which consent shall not be unreasonably withheld, conditioned or delayed if such installations, alterations, additions or improvements are reasonably consistent with the permitted use of the Premises hereunder; provided, however, that if any proposed installations, alterations, additions or improvements, in Millipore's reasonable judgment, are either unlikely to be of use to Millipore after the termination of the Lease or are likely to be expensive or difficult to remove, then Millipore may require Entegris to remove such installations, alterations, additions or improvements at the expiration or earlier termination of the Facility Term. Upon completion of any installations, alterations, additions or improvements to the Premises, Entegris shall provide Millipore with a copy of any plans and specifications (or similar drawings) prepared for Entegris in connection therewith; if Entegris is required to submit plans and specifications which have been prepared by a licensed civil engineer or architect in order to obtain a building permit for any such installations, alterations, additions or improvements, Entegris shall provide Millipore with a copy of the plans and specifications which have been stamped as appropriate by such licensed civil engineer or architect, as appropriate.
- All work on or at the Premises for which Entegris is responsible or permitted to perform under these Lease Terms shall be performed at Entegris's cost and only by contractors and subcontractors approved in writing by Millipore, which approval shall not be unreasonably withheld. Entegris shall cause all contractors

and subcontractors to procure and maintain insurance coverage naming Millipore as an additional insured against such risks, in such amounts and with such companies as Millipore may reasonably require. Entegris agrees that it will procure, at its sole cost and expense, all necessary permits before making any repairs, installations, alterations, additions, improvements or removals. Millipore agrees that it will reasonably cooperate with Entegris, at no out-of-pocket cost to Millipore in obtaining any permits, licenses or approvals related to the Premises. Entegris agrees (a) that all repairs, installations, alterations, improvements or removals done by or on behalf of Entegris (including without limitation the installation of signs) shall be done in a good and workmanlike manner in conformity with all applicable laws, ordinances and regulations of all public authorities (and Entegris shall be responsible for any additional installations or improvements required by any such laws, ordinances or regulations as a result of any repairs, installations, alterations, improvements or renewals done by or on behalf of Entegris) and all insurance inspection or rating bureaus having jurisdiction, (b) that the structure of the buildings in which the Premises are located or the operations in any other party in such buildings will not be endangered or impaired as a result of any such repairs, installations, alterations, improvements or removals done by or on behalf of Entegris, and (c) that Entegris will repair any and all damage caused by or resulting from any such repairs, installations, alterations, additions, improvements or removals, including, without limitation, the filling of holes. Entegris agrees to save Millipore harmless from, and indemnify Millipore against, any and all claims by third parties for injury, loss or damage to persons or property to the extent caused by or resulting from the doing of any such repairs, installations, alterations, improvements or removals by or on behalf of Entegris. Entegris agrees to pay promptly when due all charges for labor and materials in connection with any work done by Entegris or anyone claiming under Entegris upon the Premises so that the Premises and all such buildings shall at all times be free of liens in respect of such labor and materials. If such a lien is filed, then Entegris shall immediately, after Millipore has delivered written notice of the filing thereof to Entegris, either pay the amount of the lien or diligently contest such lien. If Entegris fails to timely take either such action, and after written notice to Entegris of its intention to do so, then Millipore may pay the amount of the lien, and any amounts so paid, including expenses and interest, shall be paid by Entegris to Millipore immediately after Millipore has invoiced Entegris therefor, and until paid, such amount shall be additional Rent.

Millipore may in its discretion and at its own cost, but shall not be obligated to, make repairs, installations, alterations, additions or improvements to the Premises at any time during the Facility Term as it determines necessary or appropriate, provided that no such work will be done if it will materially adversely impact Entegris' use of the Premises, unless Millipore will have previously made accommodations to eliminate or minimize such adverse impact and reasonable advance written notice has been given to Entegris.

74

- Entegris acknowledges that the Premises are being leased in their "AS IS" condition as of the Effective Date and that no warranties or representations, express, implied or statutory, have been made by Millipore or any agent, employee or representative of Millipore to Entegris, as to condition (environmental or otherwise), compliance with law, or fitness or suitability for any purpose other than as hereinafter set forth, all of which are expressly disclaimed. Millipore agrees that if at any time during the Facility Term the Premises are altered by an act or omission of Millipore (other than as specifically permitted in Section 5 of these Lease Terms) such that Entegris is not reasonably able to conduct such manufacturing and warehouse operations in substantially the manner in which it had been conducting such operations prior to the Effective Date, that circumstance shall constitute a material breach by Millipore of its obligations under this Agreement and that Entegris may then terminate this Agreement upon written notice to Millipore. Entegris acknowledges that except as set forth herein, Entegris has not been induced or persuaded by, nor has Entegris relied upon, any statement, promise or representation made by Millipore or any agent, employee or representative of Millipore, orally or in writing, as an inducement to entering into this Agreement including, without limitation, those relating to land use, zoning, hazardous or toxic wastes or other environmental matters.
- At the expiration or earlier termination of the Facility Term Entegris shall surrender all keys to the Premises, remove all its furnishings, fixtures, equipment (including the Entegris Equipment, subject to any provisions in this Agreement to the contrary) and other personal property now or hereafter located in the Premises, purchased or leased by Entegris with its own funds, which are not affixed to the buildings in which the Premises are located. Entegris may also so remove any furnishings, fixtures, equipment which are affixed to buildings in which the Premises are located if Millipore has agreed to such removal in writing. Entegris shall repair all damage caused by any such removal, and yield up the Premises broom-clean and in the same good order and repair in which Entegris is obliged to keep and maintain the Premises by the provisions of these Lease Terms. Any property not so removed shall be deemed abandoned and may be retained by Millipore or may be removed and disposed of by Millipore in such manner as Millipore shall determine and Entegris shall pay Millipore the directly related and reasonable cost and expense incurred by Millipore in effecting such removal and disposition and in making any incidental repairs and replacements to the Premises.
- 8. Indemnity
- 8.1 Entegris agrees that during the Facility Term and for such further time as Entegris shall occupy or hold the Premises or any part thereof, Millipore shall not be liable to Entegris and Entegris shall defend (with counsel reasonably acceptable to Millipore), indemnify, and hold Millipore, its officers, directors, agents, employees, independent contractors, and invitees harmless from and against any

and all liability for all injuries, losses, accidents or damage to any person or property, and from all claims, actions, proceedings and costs in connection therewith, including reasonable counsel fees, to the extent arising from (a) any act, omission, fault or negligence of Entegris, its officers, directors, agents, employees, independent contractors, licensees, or invitees, and relating to the presence of any of the foregoing persons or of Entegris' operations on the Premises or anywhere at the 80 Ashby Road facility.

- Millipore agrees that during the Facility Term and for such further time as Entegris shall occupy or hold the Premises or any part thereof pursuant to a reasonable claim of right, Entegris shall not be liable to Millipore and Millipore shall defend (with counsel reasonably acceptable to Entegris), indemnify, and hold Entegris, its officers, directors, agents, employees, independent contractors, and invitees harmless from and against any and all liability for all injuries, losses, accidents or damage to any person or property, and from all claims, actions, proceedings and costs in connection therewith, including reasonable counsel fees, to the extent arising from any act, omission, fault or negligence of Millipore, its officers, directors, agents, employees, independent contractors, licensees, or invitees and relating to the presence of any of the foregoing persons on the Premises or relating to Millipore's operations on the Premises or anywhere at the 80 Ashby Road facility.
- 8.3 Each of Millipore and Entegris hereby releases and waives any claim it might have against the other for any loss or damage caused by theft, destruction, fire or any other casualty, regardless of whether such loss or damage shall be brought about by the fault or negligence of the other party or its agents, to the extent the same (a) is insured against under any insurance policy that covers the Premises or the buildings in which the Premises are located, Millipore's or Entegris' fixtures, personal property, leasehold improvements or business, or (b) is required to be insured against under the terms hereof, it being understood and agreed that the waiving party reserves any rights with respect to any excess loss or injury over the amount recovered on account of such insurance. Each of Millipore and Entegris agrees to cause its insurance carrier to endorse all applicable policies waiving the carrier's rights of recovery under subrogation or otherwise against the other party.
- In addition to its obligations under Section 8.2, Millipore agrees that Entegris shall not be liable to Millipore and Millipore shall defend (with counsel reasonably acceptable to Entegris), indemnify, and hold Entegris, its officers, directors, agents, employees, independent contractors, and invitees harmless from and against any and all liability for all injuries, losses, accidents or damage to any person or property, and from all claims, actions, proceedings and costs in connection therewith, including reasonable counsel fees, arising from any release of hazardous waste by Millipore or its employees or agents in the Premises, only to the extent not caused by Entegris or its employees or agents.

- In addition to its obligations under Section 8.1, Entegris agrees that Millipore shall not be liable to Entegris and Entegris shall defend (with counsel reasonably acceptable to Millipore), indemnify, and hold Millipore, its officers, directors, agents, employees, independent contractors, and invitees harmless from and against any and all liability for all injuries, losses, accidents or damage to any person or property, and from all claims, actions, proceedings and costs in connection therewith, including reasonable counsel fees, arising from any release of hazardous waste by Entegris or its employees or agents in the Premises, only to the extent not caused by Millipore or its employees or agents.
- Insurance
- During the Facility Term, Millipore shall maintain all risk property insurance in an amount at least equal to the replacement cost of (a) the buildings in which the Premises are located and (b) Millipore property and equipment inside those buildings (excluding Entegris' personal property and any alterations, installations, additions or improvements made to such buildings by Entegris). Such insurance may be maintained under a blanket policy covering the Premises and other real estate of Millipore or its affiliated business organizations. The policies of such insurance may be payable in case of loss to the holders of any mortgages upon such buildings or the 80 Ashby Road facility as their interests may appear. Entegris shall be named as an additional insured on such Millipore property insurance, as its interests may appear. Certificates of insurance evidencing all such coverages and Entegris additional insured status for the foregoing shall be supplied to Entegris promptly upon execution of this Agreement and as such policy(ies) renew or are changed.
- During the Facility Term, Entegris will maintain at its own expense comprehensive general liability insurance, including umbrella coverage, with respect to the Entegris Equipment and its operations on the Premises, naming Millipore as additional insured, in amounts which shall, as of the Effective Date, be not less than Five Million Dollars (\$5,000,000.00) with respect to injuries to any one person and not less than Five Million Dollars (\$5,000,000.00) with respect to property damage liability. From time to time during the Facility Term, Millipore shall have the right in its reasonable discretion to require that the limits of such insurance coverage be increased, provided however, that such increases shall be in accordance with limits that are customarily carried in the metropolitan Boston area on property similar to the Premises used for similar purposes.
- 9.3 During the Facility Term, Entegris shall maintain workers compensation insurance covering all of Entegris' employees in such coverage amounts as required by law and employer's liability insurance in an amount of at least Five Hundred Thousand Dollars (\$500,000.00). During the Facility Term, Entegris shall maintain all risk property insurance in an amount at least equal to the

replacement cost of Entegris' personal property on the Premises and any other alterations, installations, additions or improvements made to the Premises by Entegris. Said replacement cost for Entegris property shall be clearly delineated and summarized promptly upon execution of this Agreement, at each renewal/change of said Entegris property policies, or at any other point of significant change in magnitude of Entegris properties on Millipore Premises. During the Facility Term Entegris shall maintain comprehensive automobile liability insurance in an amount of at least Five Hundred Thousand Dollars (\$500,000.00) per person and per accident for bodily injury, and of at least One Hundred Thousand Dollars (\$100,000) per accident for property damage. Millipore shall be named as an additional insured on the foregoing Entegris employer's liability, property, and automobile liability insurances, as its interests may appear. Certificates of insurance evidencing all such coverages and Millipore additional insured status for the foregoing and for comprehensive general liability and umbrella liability coverages shall be supplied to Millipore promptly upon execution of this Agreement and as any policies renew or are changed.

- All policies of insurance obtained by Entegris pursuant to the terms of this Lease shall be issued by insurance companies reasonably acceptable to Millipore and authorized to do business in the Commonwealth of Massachusetts. All such policies shall provide for no less than thirty (30) days prior notice to Millipore and any mortgagee of the Premises of their commencement or cancellation. Within ten (10) days after execution of this Agreement, Entegris shall deliver to Millipore certificates thereof. Entegris shall also deliver to Millipore each renewal certificate thereof as soon as is reasonably practicable after receiving request therefore by any mortgage lender of Millipore and at least fifteen (15) days prior to the expiration of the policy it renews. If Entegris fails to obtain and maintain any insurance policies required to be obtained and maintained by Entegris pursuant to Section 9.2 or 9.3 of these Lease Terms, then, after written notice to Entegris of its intention to do so and notification from Entegris of its unwillingness or inability to acquire such insurance coverages, Millipore may obtain and maintain such insurance policies (without creating any obligation on behalf of Millipore to do so), and Entegris shall pay to Millipore the cost thereof within ten (10) days after Millipore has invoiced Entegris therefor, and until paid, such amount shall be additional Rent.
- 10. Casualty or Taking
- In the event that the Premises, or more than 25% thereof, shall be taken by any public authority or for any public use, or shall be destroyed or damaged by fire or casualty, or by the action of any public authority, and the same cannot, in Millipore's reasonable judgment, be restored within 180 days from such taking or casualty, or if in Entegris' reasonable judgment, such circumstance would prevent it from continuing the conduct of its manufacturing operations at the Premises substantially in the manner contemplated under this Agreement, then this Agreement may be terminated at the election of either Millipore or Entegris. Such election shall be made by either party, by the giving of notice to the other party within thirty (30) days following such taking or casualty.

- 10.2 If neither Millipore nor Entegris exercises said election, this Agreement shall continue in force and a just proportion of the Rent reserved, as mutually agreed between the parties, according to the nature and extent of the damages sustained by the Premises, shall be suspended or abated until the Premises, or what may remain thereof, shall be put by Millipore in proper condition for use and otherwise restored to its former condition or the substantial equivalent, which Millipore covenants to do with reasonable diligence to the extent permitted by the net proceeds of insurance recovered or damages awarded for such taking, destruction or damage and subject to zoning and building laws or ordinances then in existence. "Net proceeds of insurance recovered or damages awarded" refers to the gross amount of such insurance or damages less the reasonable expenses of Millipore in connection with the collection of the same, including without limitation, fees and expenses for legal and appraisal services. If the Premises are not so restored to its former condition or the substantial equivalent within one-hundred eighty (180) days from such taking or casualty described at Section 10.1 above, this Agreement may be terminated by Entegris upon thirty (30) days prior written notice to Millipore but only if such restoration is not completed prior to the expiration of the 30 day notice period.
- 11. Defaults and Remedies
- 11.1 Events of Default. If Entegris shall default in the performance of any of its obligations to pay the Rent hereunder and if such default shall continue for thirty (30) days after written notice from Millipore designating such default, or if within thirty (30) days after written notice from Millipore to Entegris specifying any other default or defaults and Entegris has not then commenced diligently to correct the default or defaults so specified, then, and in either of such events, Millipore and the agents and servants of Millipore lawfully may, (in addition to and not in derogation of any remedies for any preceding breach of covenant), immediately or at any time thereafter and after demand and notice and without process of law enter into and upon the Premises or any part thereof or mail a notice of termination addressed to Entegris, and repossess the same as Millipore's former estate and expel Entegris and those claiming through or under Entegris and remove its and their effects without being deemed guilty of any manner of trespass and without prejudice to any remedies which might otherwise be used for arrears of rent or prior breach of covenant, and upon such entry or mailing as aforesaid this Agreement shall terminate, and Millipore, with notice to Entegris, may store Entegris's effects, and those of any person claiming through or under Entegris at the expense and risk of Entegris.
- Millipore may, but shall not be obligated to, cure, at any time, following the expiration of the applicable cure period as set forth in Section 11.1 except in cases of emergency when no notice shall be required, any material default by Entegris

Page 40 of 50

under these Lease Terms; and whenever Millipore so elects, all direct and reasonable costs and expenses incurred by Millipore, including reasonable attorneys' fees, in curing a material default shall be paid by Entegris to Millipore on demand.

- At all times during the Facility Term, Millipore reserves the right for its employees or contractors to enter the Premises as it determines necessary or desirable, for purposes of (i) ensuring compliance with this Agreement, (ii) health, safety and environmental compliance, (iii) maintenance or repair as required or permitted under this Agreement, or (iii) other reasonable purposes relating to the Premises or to any areas of its 80 Ashby Road facility other than the Premises, provided in each case that such entry into the Premises does not unreasonably interfere with Entegris' use of the Premises.

 Millipore shall provide advance notice of such entry to Entegris where reasonably practicable.
- 13. Entegris shall not, without Millipore's prior written consent, (a) assign, transfer, or encumber this Agreement or any interest therein, whether directly or by operation of law, (b) sublet any portion of the Premises; (c) grant any license, concession or other right of occupancy or any portion of the Premises; or (d) permit the use of the Premises by any parties other than Entegris. The foregoing provisions shall not apply to either (x) transactions with an entity into which Entegris is merged or consolidated or to which all or substantially all of its assets are transferred, or (y) transactions with any entity which controls Entegris or is controlled by Entegris or is under common control with Entegris.
- 14. Subject to the provisions of these Lease Terms, including but not limited to Section 5 hereof, Millipore covenants and warrants that upon performance by Entegris of its obligations under these Lease Terms, Millipore will keep and maintain Entegris in exclusive, quiet, peaceable, and uninterrupted possession of the Premises during the Facility Term.
- 15. Millipore shall at all times comply with all laws, orders, ordinances and other public requirements now or hereafter affecting the Premises.
- Any obligation of either party to indemnify the other party pursuant to these Lease Terms or this Agreement shall be barred or limited to the extent that failure of the Indemnified Party to give prompt notice to the Indemnifying Party of a claim giving rise to such indemnification obligation, or failure of the Indemnified Party to provide the Indemnifying Party with all reasonably necessary assistance and cooperation in connection therewith, prejudices the position of the Indemnifying Party in providing such indemnity hereunder. The Indemnifying Party shall have sole control of the defense with respect to any such claim (including settlement thereof). The Indemnifying Party shall keep the Indemnified Party informed at all times as to the status of the matter and the Indemnifying Party's efforts in connection therewith. The Indemnifying Party shall not settle any such claim whereby the settlement imposes upon the

Indemnified Party an obligation to take some action or refrain from doing so, without the prior written consent of the Indemnified Party, which shall not be unreasonably withheld or delayed.

17. Neither party shall unreasonably withhold or delay its consent with respect to any matter for which such party's consent is required or desirable under these Lease Terms or this Agreement.

Page 42 of 50

BEDFORD MEMBRANE MFG.			
Entegris Space Occupancy	=	6183 Sq. Ft.	
Type:	Location	Sq. Ft.	Comments
Manufacturing:			
Extrusion	D-101	1596	
	D-102	204	
	D-104	173	
HF Annealing	F-105	985	
Extraction	C-103	720	
Estatedon	F-106	323	
MFG Support	F-131	128	
Ovens	F-115*	1030	
	Total Mfg. (Sq. Ft.) = 51		
	10ttil 1411g. (5q. 1 t.)	155	
Warehouse:			
Incoming	D-108	240	15 Pallets Spaces allowance
Raw Materials	D-108	464	29 Pallet Spaces allowance
Finished Goods	D-108	128	8 Pallet Spaces (safety stock +20%)
WIP Staging	C-101	128	8 Pallet Spaces allowance
Flammable Storage	D-183/184	0	Nothing stored in Flammable storage area

Attachment A

Offices:

Bldg. F (2) 64 4 cubicles

Total Whse. (Sq. Ft.) = 960

Page 43 of 50

^{*} The parties acknowledge that the square footage for location F-115 is not precise and that certain hallway space used by Entegris may not be accounted for in this table, but for purposes of this Attachment the total square footage is a fair and accurate representation of the occupied space.

Typical Raw Mat. Storage

DOP	3
Mineral Oil	2
PE	2
Base-30	12
Blue Patapar	1
Genesolve	3
Vertrel	3
Cores	1
Cartons/Boxes	2
	29

[ATTACH DRAWINGS OF PREMISES FROM OLD AGREEMENT]

FILM 1 Annealing Line

where N = Number of rolls in the job

L = Target length of end roll in feet

P = Extra material run to get L in feet

1.05 = Set-up time factor

S = Machine speed in feet per minute

60 = Conversion from minutes to hours

Page 44 of 50

EXHIBIT C

The Supplement

ARTICLE I – MEMBRANE MANUFACTURING OPERATIONS L.1. Film 1 Manufacturing Operations

- **I.1.1.** Millipore retains ownership of the Millipore Equipment including the related processes (financially and operationally). Entegris retains ownership of the Entegris Equipment including the related processes (financially and operationally).
- **I.1.2.** The Film 1 Annealing Line manufacturing process will be staffed by regular Millipore employees when possible and supervised / managed by Millipore. Only regular Millipore employees certified on Film I operations will be rotated into Film I manufacturing operations. Exceptions may include new or contract employees under supervision of an employee certified for Film 1 operation.
- **I.1.3.** Entegris is responsible for technical direction as to all process operating conditions (a la Millipore Product Owner model) for UPE Film 1 manufacturing activities.
 - **I.1.4.** Entegris is responsible for UPE process development.
 - I.1.5. Entegris is responsible for documentation and training for changes to the job content for UPE processing on Film I.
 - I.1.6. Millipore is responsible for maintaining and repairing Film 1 manufacturing equipment to be in good and efficient working order.
- **I.1.7.** Millipore proposed modifications for non-Entegris UPE membranes that could influence UPE running on Film I process will be subject to mutual agreement.
- **I.1.8.** Priorities during the time when UPE is running on the Film 1 Annealing Line shall be established by Entegris, subject to Millipore's recommendations for run priorities for non-Entegris UPE Membranes.
- **I.1.9.** Millipore will give Entegris advance notice of the schedule for pending runs of non-Entegris UPE membranes on Film 1 Annealing Line. This notice would consist of a quarterly schedule with monthly updates.
- **I.1.10.** All of the foregoing obligations of both parties under this Section I.1 shall expire on the earlier of (i) the expiration of the Facility Term, or (ii) the date Entegris is no longer operating its Film 1 Annealing Line at the Premises.

Page 45 of 50

I.2. VMF 4 Manufacturing Operations

- **I.2.1.** Entegris will submit its estimated VMF 4 total quarterly requirements prior to the beginning of each quarter including both Entegris production requirements and R&D requirements.
- **I.2.2.** These requirements will be translated by Millipore into an adequate number of production days to meet the quarter's estimated requirements. These days will be identified in a quarterly schedule by Millipore which shall be routinely available to Entegris. Schedule changes during the quarter will be reflected in an update to the quarterly schedule and communicated to Entegris in a timely fashion.
- **I.2.3.** Entegris shall have control of the detailed schedule for operations during these Entegris dedicated days and will, in-turn, provide Millipore with the detailed schedule for the Entegris dedicated days at least one week in advance.
- **I.2.4** In the event of a high priority urgent need combined with no Entegris time scheduled within an upcoming two week window, Entegris shall be entitled to a 24 hour emergency run on the VMF 4 equipment within the following week. This right is intended only in case of emergency and it is anticipated that this set of circumstances would only come up during times of low Entegris volume on VMF 4.
- **I.2.5.** Entegris shall be entitled to propose corrections to the dedicated operating schedule for the quarter to Millipore; subject to mutual agreement, the schedule of Entegris dedicated days will be revised accordingly for the remainder of the quarter.
- **I.2.6.** Detailed plans for periods of Entegris R&D VMF 4 operations will be submitted in advance to Millipore. Where proposed R&D operations involve extensions of existing chemistries on UPE substrates, this pre-notification would need to be at least one week in advance and will be subject to Millipore's approval. Where proposed R&D operations involve new chemistries or new substrates, Entegris shall advise Millipore of that fact and shall submit any details that Millipore reasonably requests and will be subject to Millipore's approval within one month following the submission. In the event that Millipore does not give its approval, it shall advise Entegris as to the reasons why the proposal was not acceptable in reasonable detail and shall consult with Entegris as to alternatives that might be acceptable.
- **I.2.7.** Millipore will continue to be responsible for planning, procurement and performance of incoming QC of all standard chemicals in support of the Entegris dedicated VMF 4 operating schedule for each quarter. Any non-standard chemicals required for Entegris R&D operations will be provided by Entegris R&D.
- **I.2.8.** Entegris documentation shall govern production of all Entegris RTM'd products. With consultation support from Millipore as may be required, Entegris shall also be responsible for the upgrade and updating of all Entegris documentation and for notification and training of the Millipore Supervisor and Product Owner as designated by Millipore.

- **I.2.9.** VMF 4 operations shall be staffed by regular Millipore employees where possible and supervised /managed by qualified Millipore personnel. VMF 4 operations will always be staffed by at least one employee certified in the operation of VMF 4.
 - I.2.10. Entegris shall be responsible for UPE process development with respect to hydrophilic membranes.
 - I.2.11. Millipore shall be responsible for maintaining and repairing all VMF 4 equipment in good and efficient working order.
- **I.2.12.** Millipore shall be responsible for supervision, technical direction and support during the running of standard Entegris production materials and this routine supervision and support, as well as all other services provided pursuant to this Agreement in connection with Membrane manufacture, is included within the rate structure specified in Section 7.1 of the Agreement. To the extent Millipore provides extraordinary services to Entegris, such services will be charged in accordance with the rate structure specified in Section 7.3 of the Agreement.
 - **I.2.13.** Identified Entegris contact(s) must be available 24 hours per day for technical consultation.
- **I.2.14.** Millipore shall have approval authority with respect to any Entegris proposed modifications to VMF 4 equipment for UPE membrane processing. Millipore agrees that proposed modifications to the VMF 4 equipment for non-Entegris UPE membrane processing that could influence the processing of UPE membranes on the VMF 4 equipment shall be reviewed with Entegris and shall be subject to mutual agreement.
- **I.2.15.** Entegris may shift production of its membranes to its own hydrophilization machine (to be located at a Entegris location). Any such shifting of production shall not result in a termination of the Agreement, which may be only be terminated in accordance with its terms.

I.3. Minimum Lot Size and Minimum Scrap levels for VMF 4 and Film I Production

- **I.3.1.** Intentionally omitted.
- **I.3.2.** Entegris shall not be responsible for time & materials for unsuccessful VMF 4 campaigns > 3 rolls in duration where phobic membrane is determined not to be the cause, unless Entegris authorizes extending the run beyond three rolls in which case Entegris will document that authorization in the form of an email or letter.
- **I.3.3.** Entegris shall not be responsible for the cost of time and materials incurred in unsuccessful VMF 4 process runs where it is demonstrated that Millipore failed to adhere to the process documentation and to Entegris's technical directions.

Page 47 of 50

I.3.4. Entegris shall not be responsible for the cost of time and materials incurred in unsuccessful Film 1 process runs where it is demonstrated that Millipore failed to adhere to the process documentation and to Entegris's technical directions.

I.4. Revised Responsibilities for Regulatory & HSE Compliance

I.4.1 Responsibilities.

Current responsibilities shall be revised to reflect current operating procedures, including: Entegris has assumed full responsibility for waste disposal in connection with waste generated by the Entegris Equipment. Millipore continues to provide spill control support to ensure prompt response to any emergency situation, however, Entegris is responsible for any follow-up actions that are required as a result of any emergency situation. VMF 4 waste treatment costs continue to be included in VMF 4 rates.

Entegris is solely responsible for the generation, control, compliance and reporting of chemical emissions resulting from its dedicated operations located at the Millipore Bedford site. Millipore is responsible for the generation, control, compliance and reporting of chemical emissions from its operations. Where Entegris products are processed on Millipore Equipment, Entegris and Millipore will work together to ensure compliance with federal, state and local regulations, and conformance with EHS self certifications programs similar to ISO 14001. Furthermore, as dedicated Entegris-owned operations may have a potential impact on the surrounding community, Entegris agrees to continue to make reasonable commercial attempts to lessen the environmental impact of its operations at the Bedford site and to notify Millipore of any chemical spills, notices of non-compliance or inspections.

I.4.2 Incident Response.

Entegris shall notify Millipore of all spills or releases greater than the Reportable Quantity, as defined by the State of Massachusetts and Town of Bedford, whichever is less. Entegris is responsible for timely notification and applicable follow up of regulatory agencies, as required by law or best practice. For any spill or release, Entegris shall ensure a clean up scheme that follows all of the requirements set forth in OSHA 29 CFR 1910.120 - Hazardous Waste Operation and Emergency Response, and the Massachusetts Contingency Plan, 310 CMR 40. In the event of a spill or release from an Entegris process requiring a clean up, Millipore shall provide Entegris with clean-up services using Millipore Corporation's Emergency Response Team. In the event of a spill or release from an Entegris process requiring a clean up using Millipore's Emergency Response Team, Millipore will conduct any cleanup services with the same degree of care it would use to respond to a spill or release from a Millipore process and shall be compensated by Entegris at the rates set forth on Annex I, subject to adjustments pursuant to Section 7.3; provided, that Entegris will compensate Millipore for at least 4 hours in the event of any such incident.

Page 48 of 50

Entegris shall also compensate Millipore for all reasonable and customary costs associated with engaging professional consulting services, as deemed appropriate by Millipore's corporate EHS department and approved by Entegris' corporate EHS department (such approval not to be unreasonably withheld or delayed, provided that if Millipore reasonably deems the need for such consulting services urgent, such as an emergency response, and cannot reach an Entegris representative in a timely manner after diligent efforts to obtain Entegris' approval, Millipore may proceed on a limited basis to address such urgent situation without Entegris' prior approval but shall notify Entegris promptly) as a result of the aforementioned spills or releases. Entegris shall also reimburse Millipore for all supplies consumed during the response event. Entegris shall be responsible for appropriate disposal of wastes associated with any clean-up.

Entegris may also contract with a third party that meets the requirements of OSHA 29 CFR 1910.120 - Hazardous Waste Operation and Emergency Response in lieu of using Millipore's Emergency Response Team.

I.4.3 Equipment Operability and Risk Mitigation.

Entegris shall perform an initial process hazard analysis (hazard evaluation), and proactively manage any subsequent changes, on any process or equipment ("Processes") proposed for installation or use on Millipore managed property. The process hazard analysis shall be completed with current process safety information and be reasonable with respect to the complexity of the Processes and shall identify, evaluate, and mitigate hazards involved in the Processes. Entegris shall conduct and implement the process hazard analysis and a management of change procedure in a manner consistent with OSHA 29 CFR 1910.119 - Process Safety Management and in conjunction and consultation with Millipore's EHS staff. Entegris shall implement all reasonable corrective actions or improvements identified by the process hazard analysis in a timely manner and at its sole expense, in all cases in accordance with applicable regulatory and reasonable Millipore requirements, unless such corrective actions are facility related, in which case such actions will be at the expense of Millipore to the extent not caused by Entegris. Copies of the process hazard analysis, corrective actions and implementation schedules shall be provided to Millipore promptly upon availability. Furthermore, all applicable process safety information will be updated with any change to the process not considered a replacement in kind.

I.5. Entegris Option to Assume Supply Chain Responsibilities

I.5.1. During the Facility Term, upon six months prior notice to Millipore, Entegris shall have the option to assume responsibility for planning and procurement for all raw materials used in the manufacture of UPE Membranes and Treated Other Entegris Membranes. If Entegris assumes this responsibility, Entegris shall be accountable for un-utilized machine time occurring as a result of raw material availability if Millipore cannot utilize that machine time for any

Page 49 of 50

alternative membranes. After the Facility Term, Entegris will have all, and Millipore will have no, responsibility for planning and procurement of any raw materials (other than with respect to VMF4) unless the parties otherwise agree in writing.

I.5.2. Notwithstanding the exercise by Entegris of the option in I.5.1, Entegris shall continue to be entitled to contract with Millipore for incoming quality control services.

I.6. Resolution of Technical Issues.

- **I.6.1.** In the event of chronic low yielding runs, Entegris can issue a corrective action request. This request would result in a inter-company meeting(s) to develop a mutually agreed upon corrective action plan within two weeks of the issuance of the request.
- **I.6.2.** Millipore will support this activity with at least one mutually agreed upon representative (usually the Production Supervisor and / or the Millipore Product Owner).
- **I.6.3.** In the event that this vehicle is unsuccessful in resolving the issue, the corrective action team will be supported by additional mutually agreed upon representatives with special expertise from both/either Millipore and Entegris. The cost of Millipore support for this "above and beyond" technical support will be at the technical support rate established in accordance with Section 7.3 of the Agreement.

Page 50 of 50

*** AMENDED - SEE ADDENDUM***

Fluoropolymer Purchase and Supply Agreement

This Agreement, by and between

E. I. DU PONT DE NEMOURS AND COMPANY, a corporation of the State of Delaware having offices at 1007 Market Street, Wilmington, Delaware 19898 (hereafter "DuPont") and

ENTEGRIS INC., a corporation of the State of Delaware having offices at 129 Concord Road, Billerica, Massachusetts 01821 (hereafter "Entegris")

WITNESETH:

WHEREAS DuPont manufactures and sells fluoropolymer resins;

WHEREAS Entegris has for some period of time purchased fluoropolymer resins from DuPont; and

WHEREAS DuPont desires to continue to supply fluoropolymer resins to Entegris, and Entegris desires to continue to purchase such resins from DuPont;

NOW, THEREFORE, in consideration of the mutual covenants and promises set forth herein, the parties agree as follows:

ARTICLE 1 TERM AND SCOPE

- 1.1 The terms of this Agreement shall apply to supply by DuPont and its subsidiaries and affiliates and purchases by Entegris and its subsidiaries and affiliates of the products described in Article 2. The subsidiaries and affiliates of DuPont and Entegris are listed on Schedule 1 attached hereto and incorporated herein by reference. This Agreement shall cover the period January 1, 2010 to December 31, 2012. No less than six (6) months prior to the end of this Agreement, the parties Will enter into discussions regarding the terms and conditions under which this Agreement can be extended.
- 1.2 This Agreement shall relate solely to the following matters:
 - (A) supply of PFA resins by DuPont to Entegris in all regions where Entegris uses PFA to manufacture parts and components;
 - (B) supply of PTFE resins by DuPont to Entegris in all regions where Entegris uses PTFE to manufacture parts and components;
 - (C) supply of PFA film by DuPont to Entegris in all regions where Entegris uses PFA film to manufacture parts and components; and

(D) Supply of ETFE resins by DuPont in all regions where Entegris uses ETFE to manufacture parts and components.

ARTICLE 2 PRODUCT

2.1 DuPont will sell, and Entegris will purchase the following PFA resins and films ("PFA Product"), ETFE resins ("ETFE Product") and PTFE resin ("PTFE Product") as further described in the Entegris Resin Specification & Control Plans and Entegris Source Control Specifications that are mutually agreed to:

Entegris Part No.	DuPont Code
105	ETFE210
191	ETFE2181
192	ETFE HT2185
U33494	Granular 7A
1114-009	PFA 2000LP film
115	PFA 440HP-A
158	PFA 440HP-B
132	PFA 440HP-D
164	PFA 445HP
146	PFA450HP
01-1009836	PFA 450HP (diaphragms)
1005	PFA 940HP Plus
1006	PFA 950HP Plus
01-1004803	PFA C5260
160	PFA TE5789
103-F	PFA TE9738J
	PFA TE9738JN
100000	PFA 340J
102300	PFA 420HPJ
102100	PFA 440HPJ
102600	PFA 451HPJ

New resins introduced into the market by DuPont are not included as part of this agreement, but can be made part of this Agreement upon mutual agreement between Entegris and DuPont in accordance with the procedures described in paragraph 15.1 below.

ARTICLE 3 QUALITY ASSURANCE

3.1 DuPont warrants to Entegris that the PFA Product, ETFE Product and PTFE Product will conform to DuPont's specifications and DuPont will provide certification with each shipment of PFA Product, ETFE Product, and PTFE Product indicating compliance with appropriate Entegris Resin Specifications. Other than the foregoing, DUPONT MAKES NO WARRANTIES OF MERCHANTABILITY OR OF FITNESS FOR A PARTICULAR

PURPOSE EVEN IF THAT PURPOSE IS KNOWN TO DUPONT, NOR ANY OTHER EXPRESS OR IMPLIED WARRANTY. Entegris assumes all risk and liability for results obtained by the use of the PFA Product, ETFE Product, and PTFE Product covered by this Agreement, whether used singly or in combination with other products.

- 3.2 DuPont agrees to promptly notify Entegris, in writing, of any anticipated manufacturing, raw material or process changes affecting form, fit or function of the PFA Product, ETFE Product, and PTFE Product a full sixty (60) days prior to the execution of such a change.
- Further, DuPont agrees to notify Entegris of any intent to discontinue any of the Products named in Section 2.1, or other Products that may be added to Section 2.1 in the future at least six (6) months in advance. In the event of such discontinuance, Entegris shall have the right to purchase a two (2) year supply of the Product(s) at the then current price (the "Final Purchase Option") provided that Entegris gives DuPont written notice of its intent to exercise the Final Purchase Option within ninety (90) days of DuPont's notice of intent to discontinue the Product(s). The maximum quantity of the Final Purchase Option for each Product shall be twice the prior twelve (12) month purchase amount. At DuPont's sole discretion, the Final Purchase Option may be for larger quantities of Product. Entegris agrees to take delivery of the final purchase by the end of the calendar year in which the order is placed. DuPont agrees to hold in inventory such additional inventory for a period not to exceed six (6) months at no charge to Entegris and material can not cross the end of a calendar year.

ARTICLE 4 INSPECTION

4.1 For the purpose of confirming compliance with Entegris Resin Specifications, all PFA Product, ETFE Product, and PTFE Product supplied hereunder shall be subject to inspection by Entegris. In the event that Entegris and DuPont mutually agree that any of the PFA Product, ETFE Product, or PTFE Product shipped to Entegris under this Agreement does not meet the appropriate Entegris Resin Specifications, then Entegris shall have the right to reject such Product by giving DuPont prompt notice thereof. DuPont may, at its option, obtain samples of the rejected PFA Product, ETFE Product, or PTFE Product from Entegris for analysis. DuPont will, subject to availability, endeavor to ship replacement PFA, ETFE, or PTFE Product to Entegris within seven (7) days of receipt of notification of such rejection and will fully credit Entegris for the rejected PFA, ETFE, or PTFE Product upon receipt of such rejected Product. DuPont will be responsible for any expenses incurred in returning such rejected Product.

ARTICLE 5 PACKAGING

5.1 PFA Product will be delivered in returnable bulk containers (each nominally holding eighteen hundred (1800) pounds), or twenty-five (25) kilogram bags with an inner bag constructed of low density polyethylene liner and an outer bag constructed of kraft paper, or other containers acceptable to DuPont subject to the prior written approval of Entegris. PTFE Product will be delivered in one hundred (100) pound plastic drums with

polyethylene liners. ETFE Product will be delivered in forty-five (45) kilogram bags. Product in Japan will be delivered in twenty-five (25) kilogram bags. Returnable bulk containers will be delivered to U.S. Chaska location only.

ARTICLE 6 QUANTITY

- Entegris commits that during the term of this Agreement, Entegris and its subsidiaries and affiliates (as listed in Schedule 1) will purchase from DuPont not less than ninety percent (90%) of Entegris' and its subsidiaries and affiliates total PFA, ETFE, and PTFE products purchase requirements for the manufacture by Entegris of parts and components. It is the expectation that in no Entegris business will DuPont's share be less than ninety percent (90%) for the products listed above. The volume of PFA, ETFE, and PTFE Products which Entegris and its subsidiaries and affiliates expect to purchase from DuPont each year of this Agreement is set forth by Entegris providing a forecast, herein known as THE ANNUAL FORECAST, which shall be provided to DuPont no later than ninety (90) days prior to the end of each calendar year. Notwithstanding the foregoing, if purchase of Products by Entegris is constrained by limitations in supply from DuPont for any reason (including Force Majeure or allocation), the parties expressly agree that Entegris' purchase commitment set forth in 6.1 shall be waived to the extent of the DuPont supply limitations.
- In calendar years 2010 through year 2012, DuPont and its subsidiaries and affiliates (as listed in Schedule 1) agree to make available for sale to Entegris and its subsidiaries and affiliates one hundred thirty percent (130%) of the volume of PFA, ETFE, and PTFE Products shown in THE ANNUAL FORECAST. DuPont upon reasonable notice will use best efforts to supply volumes of Product exceeding one hundred thirty percent (130%) of the ANNUAL FORECAST. Products sold to Entegris shall be for internal consumption only, not resell. Provided that Entegris' total requirements for PFA, ETFE, and PTFE Product equal or exceed the volume shown in THE ANNUAL FORECAST for the above-identified fiscal years, Entegris agrees to purchase said minimum volume. In the event that Entegris' total requirements for PFA, ETFE, and PTFE Products fall below the volume set forth in THE ANNUAL FORECAST for any of the years 2010, 2011 and 2012, Entegris agrees that pursuant to the terms of Article 6.1, purchases of PFA, ETFE, and PTFE Product will reflect a minimum of ninety percent (90%) of its total purchase requirements of PFA, ETFE, and PTFE products in each such year. Notwithstanding the foregoing, if purchase of Products by Entegris is constrained by limitations in supply from DuPont for any reason (including Force Majeure or allocation), the parties expressly agree that Entegris' purchase commitment set forth in 6.2 shall be waived to the extent of the DuPont supply limitations.
- 6.3 The maximum monthly quantity ordered by Entegris may not exceed one hundred ten percent (110%) of Entegris' monthly forecast, as provided for under Article 7, unless agreed to by the parties sixty (60) days prior to the requested ship date.
- The requirements of this Article 6 shall apply to all current Entegris locations and locations acquired during the Term of this Agreement which consume PFA, ETFE, and PTFE products.

ARTICLE 7 FORECAST REQUIREMENTS

- 7.1 Entegris will provide DuPont a written forecast of its requirements for PFA Product, ETFE, Product and PTFE Product on an annual basis (by Entegris part number) no later than ninety (90) days prior to the end of each calendar year (THE FORECAST YEAR).
- 7.2 Monthly forecast updates will be provided on or before the 30th of each month. This monthly forecast update will reconfirm the next month's forecast (Month #1) and will forecast quantities for the next three months (Months #2, #3, and #4).

ARTICLE 8 PURCHASE ORDERS

Valid Purchase orders from Entegris location will be issued for each PFA Product, ETFE Product, and PTFE Product specifying desired quantities by grade at least thirty (30) days in advance of the requested delivery date. Orders shall be firm and irrevocable fourteen (14) days prior to the requested delivery date. In exceptional cases, DuPont will make reasonable effort to accommodate requests to increase the volume of an order after the fourteen (14) day firm date.

ARTICLE 9 PRICE

- 9.1 The price of PFA, ETFE, and PTFE Products will be as set forth in Schedule 2, Pricing, attached hereto and incorporated by reference.
- 9.2 Pricing for new resins developed by DuPont will be agreed upon separately and incorporated as part of this Agreement should a need arise. Any such changes will be in writing and approved by both DuPont and Entegris.
- 9.3 In each country where DuPont sells PFA Product to Entegris, should DuPont sell to a third party PFA Product, or any other first quality Teflon® PFA HP or HPPLUS resin, in equal or less volume, suitable for use in applications and markets similar to Entegris' current applications and markets at a price lower than the price of PFA Product sold to Entegris hereunder, then DuPont shall offer Entegris the same PFA resin at the lower price for such term as the lower price is in effect with the third party to the extent DuPont is permitted to offer such lower price by applicable laws and regulations.
- 9.4 U.S. payment terms will be thirty (30) days from receipt of invoice by Entegris with a consolidated invoice issued for all shipments at the end of each month. For Malaysia and Korea, payment terms will be thirty (30) days from receipt of invoice by Entegris. For Japan, payment terms will be forty-five (45) days from receipt of invoice by Entegris.
- 9.5 The parties shall conduct an annual review of pricing no later than September 1st of each calendar year. Failure to reach agreement on pricing for the next calendar year shall be

grounds for termination of this Agreement by either party. Termination will be effective six (6) months after receipt of notice to terminate. In the case of termination, pricing shall remain valid as set forth herein until the Agreement terminates.

9.6 In addition to the annual price review defined in Article 9.5, DuPont agrees to engage in pricing discussions with Entegris at any time to address specific competitive situations that may require adjustment in pricing from this Agreement. It is understood that this will be done only to address competitive downstream market opportunities with non-DuPont resins. If the parties cannot reach mutual agreement on pricing, then either party may terminate this Agreement effective six (6) months after receipt of notice of intent to terminate. In the case of termination, pricing shall remain valid as set forth herein until the Agreement terminates.

ARTICLE 10 MARKETING

- 10.1 During the term of this Agreement, Entegris agrees not to actively promote any other manufacturer of PFA fluoropolymers in the marketplace.
- DuPont recognizes the contribution Entegris has made in marketing and promoting the attributes and value of Teflon® PFA HP, Teflon® PTFE and Teflon® PFA HP PLUS products to the semiconductor industry. Entegris' market information access and recognition as a leader in the industry are valued. As part of this Agreement, DuPont expects Entegris' continuing effort in gaining access and information regarding new applications and market trends.
- 10.3 The Parties acknowledge that TEFLON® is a registered trademark of DuPont for its brand of fluoropolymer resins which can only be licensed by DuPont for use in approved applications. Use of the TEFLON® trademark in connection with DuPont products is not permitted without a license. This license will be provided under separate agreement.

ARTICLE 11 SUPPLY ASSURANCE

- 11.1 Except for adverse circumstances as described in Paragraph 3 of the Standard Conditions of Sale (attached hereto as Schedule 3 and made part hereof), in the event of a production disruption impacting DuPont's ability to supply PFA, ETFE or PTFE Product hereunder, Entegris' pro rata allocation of available Product quantities will be the most favorable of any offered to any of DuPont's customers.
- 11.2 DuPont and Entegris agree to establish mutually acceptable supply chain and product storage processes that will ultimately reduce lead times and inventory.

ARTICLE 12 INDEMNITY

- DuPont shall indemnify and hold Entegris harmless from any costs, expenses (including attorneys' fees incurred that are reasonable and customary for commensurate type of litigation), losses, damages, or liabilities incurred because of actual or alleged infringement of any patent, trade secret or trademark right arising out of the use or sale by Entegris made known to DuPont as supplier of Products irrespective of whether Entegris furnishes any specifications to DuPont, except as otherwise provided in this paragraph. Entegris shall notify DuPont of such claim or demand within ten (10) days of receipt. Upon written notice of such claim or demand, DuPont shall provide defense of the claim or demand or settlement thereof. If an injunction issues as a result of any claim or action, DuPont agrees at its expense and Entegris' option to either: (i) replace Product with non-infringing products if reasonably available; or (ii) refund to Entegris the amount paid for any Product returned to DuPont or destroyed. Further, and in addition to DuPont's above described obligations, regardless of which of the foregoing remedies is effected, DuPont shall pay to Entegris rework expenses and incremental costs incurred by Entegris to procure alternative products required to fill orders placed by Entegris and accepted by DuPont as of the effective date of the injunction not to exceed a maximum value of \$1 million dollars. This indemnification shall not apply to the extent custom Products are manufactured to Entegris' detailed specifications and such infringement would not have occurred but for complying with such detailed specifications.
- 12.2 Entegris shall defend, indemnify, and hold DuPont harmless from any costs or expenses arising from an actual or alleged claim of infringement by a third party, where Entegris furnishes and requires DuPont to use detailed specifications for the process of manufacturing the Product(s), and such infringement claim would not have occurred but for complying with such detailed specifications. Notwithstanding the foregoing, Entegris shall have no liability to DuPont hereunder, if DuPont knows that the required manufacturing specification(s) infringes another party's intellectual property rights.
- 12.3 The foregoing provisions of Article 12 state the entire set of obligations and remedies flowing between DuPont and Entegris arising from any intellectual property claim by a third party.

ARTICLE 13 REPORTING

On a quarterly basis, Entegris will provide DuPont with Quality and Deliverance Performance Ratings. DuPont agrees to develop corrective action plans for review and approval by Entegris when Quality Performance Ratings fall below one hundred percent (100%) and/or Delivery Performance falls below ninety-eight percent (98%).

ARTICLE 14 CONFIDENTIALITY

14.1 The terms of the Confidentiality Agreement between the parties with effective date of April 16, 2007, shall apply to this Agreement.

ARTICLE 15 NOTICE

All notices required under this Agreement shall be in writing and shall be effective if delivered to the party entitled to receive the same by hand (or courier) to the person specified for the receiving party at the address specified or mailed (by registered or certified mail, return receipt requested) to the person specified for the receiving party. Notices required or contemplated under this Agreement shall be addressed as follows:

For DuPont:

Attention:

Sales Manager, Fluoropolymers E.I. du Pont de Nemours and Company P.O. Box 90702 Wilmington, DE 19880 Fax (302) 999-3780

For Entegris:

Attention: General Counsel Entegris, Inc. 129 Concord Road

Billerica, MA 01821

With copy to: Norman Jaillet Entegris, Inc. 129 Concord Road Billerica, Ma 01821

ARTICLE 16 MISCELLANEOUS

- 16.1 Except as otherwise specifically provided in this Agreement, the provisions of the Standard Conditions of Sale (Schedule 3) shall govern each sale and shipment made hereunder.
- 16.2 This Agreement is not assignable or transferable by either party, in whole or in part, except with the prior written consent of the other party.

- 16.3 This Agreement shall be governed by and interpreted in accordance with the laws of the State of Delaware. The courts of the State of Delaware shall have exclusive jurisdiction over any dispute relating to the terms and conditions of this Agreement.
- 16.4 With the exception of the Consignment Agreement with effective date of January 1, 2008, and the Confidentiality Agreement with effective date of April 16, 2007, this Agreement embodies the entire agreement and understanding between DuPont and Entegris relative to the subject matter hereof and there are no understandings, agreements, conditions or representations, oral or written, expressed or implied, with reference to the subject matter hereof that are not merged or superseded hereby. No amendment, modification or release from any provision hereof shall be of any force or effect unless it is in writing, signed by the party claimed to be bound thereby, and specifically refers to this Agreement.

IN WITNESS WHEREOF, the parties have caused this Agreement to be executed by their duly authorized representatives on the dates indicated below.

E. I. DU PONT DE NEMOURS AND COMPANY		ENTEGRIS, INC.	
BY:	/s/ Gary W. Spitzer.	BY	/s/ Bertrand Loy.
_	Gary W. Spitzer		Bertrand Loy
	President DC&F	Executi	ve Vice President and COO

Date: June 1, 2010

Date: May 21, 2010

ADDENDUM

The following is an Addendum (the "Addendum") to that certain Supply Agreement (the" Agreement") between **E.I. du Pont de Nemours and Company** ("DuPont") and **Entegris Inc.** ("Entegris") with effective date of January 1, 2010. Capitalized terms used in this Addendum not otherwise defined herein shall have the definitions specified in the Agreement. Effective January 1, 2011, the terms of this Addendum shall govern and control in case of conflict with the Agreement. Subject to the alterations and amendments contained in this Addendum, the parties ratify and confirm the Agreement in all other respects.

1. Article 6.1, Quantity, is deleted and replaced with the following:

Entegris commits that during the term of this Agreement, Entegris and those of its subsidiaries and affiliates as listed on Schedule 1 will purchase from DuPont not less than seventy percent (70%) of their subsidiaries and affiliates listed on Schedule 1 its total PFA, ETFE, and PTFE requirements in connection with the manufacture by Entegris of parts and components. The volume of PFA, ETFE, and PTFE Products which Entegris and those of its subsidiaries and affiliates listed on Schedule 1 expect to purchase from DuPont each year of this Agreement is set forth by Entegris providing a non-binding forecast, herein known as THE NON-BINDING ANNUAL FORECAST, which shall be provided to DuPont no later than ninety (90) days prior to the end of each calendar year. Notwithstanding the foregoing, if purchase of Product by Entegris is constrained by limitations in supply from DuPont for any reason (including Force Majeure or allocation), the parties expressly agree that the purchase commitment set forth in Article 6.1 shall be waived to the extent of the DuPont supply limitations.

2. Article 6.2, Quantity, is deleted and replaced with the following:

In calendar year 2011, DuPont and its subsidiaries and affiliates as listed on **Schedule 1** agree to make available for sale to Entegris and its subsidiaries and affiliates PFA, ETFE, and PTFE Products in volumes that exceed 2010 volumes which have been purchased and paid for by fifteen percent (15%). Products sold to Entegris shall be for internal consumption only, not resell. Quantities of Products to be supplied by DuPont to Entegris in calendar year 2012 shall be as mutually agreed upon by the parties no later than December 1, 2011.

Page 1 of 2

IN WITNESS WHEREOF, the parties have signed this Addendum, dated and effective January 1, 2010.				
E.I. du Pont de Nemours and Company		Entegris Inc.		
By:	/s/ Gary W. Spitzer.	/s/ Gideon Argov.		
Name:	Gary W. Spitzer.	Name: Gideon Argov.		
Title:	President DC+F.	Title: President & CEO.		

 $3. \ Schedule \ 2-Pricing \ is \ deleted \ and \ replaced \ with \ the \ following \ revised \ pricing:$

CERTIFICATIONS

- I, Gideon Argov, certify that:
- 1. I have reviewed this Report on Form 10-Q of Entegris, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects, the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: April 21, 2011

/s/ Gideon Argov

Gideon Argov Chief Executive Officer (Principal Executive Officer)

CERTIFICATIONS

- I, Gregory B. Graves, certify that:
- 1. I have reviewed this Report on Form 10-Q of Entegris, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects, the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: April 21, 2011

/s/ Gregory B. Graves

Gregory B. Graves Chief Financial Officer (Principal Financial Officer)

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report on Form 10-Q (the "Report") of Entegris, Inc, a Delaware corporation (the "Company"), for the period ended April 2, 2011 as filed with the Securities and Exchange Commission on the date hereof, Gideon Argov, President and Chief Executive Officer of the Company and Gregory B. Graves, Chief Financial Officer of the Company, each hereby certify pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 that:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: April 21, 2011

/s/ Gideon Argov

Gideon Argov

Chief Executive Officer

/s/ Gregory B. Graves

Gregory B. Graves Chief Financial Officer